



2015, VOL 38

Antoine Ghauch

Iron-based metallic systems: an excellent choice for sustainable water treatment

80 pages, 4 figures, 1 table, 328 references



This work is dedicated to my parents Angele and Cmdt Rajaa, my brothers Marwan and Joseph, my lovely wife Alice, my beloved children Anthony and Marc for their endless love

Preface

The present work was presented as thesis for the Habilitation degree in Chemistry at Joseph Fourier University of Grenoble 1 (France) under "Chimie et Sciences du Vivant" Doctoral School and was accepted on December 13, 2013.

This thesis investigates the important role that metallic iron (Fe^0) or Fe^0 -based systems (including bimetallics, trimetallics and nano-Fe⁰) can play in sustaining the efficiency of physical-chemical treatment systems for water decontamination. Relevant water treatment technologies include:

- Fe^0/H_2O systems using mm, μm and nm particles
- $Fe^0/Me^0/H_2O$ systems where Me^0 is a transition metal e.g. Pd, Cu, Co, Ni, etc.
- Fe^0/H_2O_2 systems known by the Fenton and Fenton-like reactions (silent system)
- $Fe^0/US/H_2O_2$ systems (sonicated system at low frequency)
- Fe⁰/Persulfate and Fe⁰/Me⁰/Persulfate systems in addition to persulfate heated systems.

Cleaning water from common and emergent contaminants becomes a priority since provision in drinking water of an increasing worldwide population continue to grow in a complicated and uncertain future in terms of global warming and environmental challenges. Directions toward cost effective, affordable and sustainable technologies for water treatment could be achieved by using cost-effective materials like Fe⁰-based alloys in combination with adequate harmless reagents.

Chapter 1 presents the scientific basis of iron corrosion and filtration on iron filters. It also addresses some unanswered questions regarding the removal mechanism of organic contaminants in Fe^0/H_2O systems. Chapter 2 presents the attached articles in relation to their contribution in order to better understand the removal mechanism of organic contaminants in Fe^0 systems (micrometric and nanometric scale) with application to antibiotics and emphasis on permeability loss. Chapter 3 presents the appended articles in the perspective of using Fe^0 amended systems (e.g. bimetallics and trimetallics) in addition to technical issues relative to experimental design with application to emergent contaminants (Pharmaceuticals and Personal Care Products, PPCPs). Chapter 4 investigates the role that nanometric and micrometric Fe^0 can play in improving the Fenton process in terms of sustainability in sonicated and silent systems. Chapter 5 shows that Fe^0 and Fe^0 -based systems (bimetallics and trimetallics) could be considered as reliable catalysts for persulfate activation yielding complete degradation of PPCPs toward full mineralization. Thermal activation of persulfate is considered as well showing the possibility of application of such advanced oxidation process to hot spot effluents. An outlook is given in form of specific recommendations for future works. This acclaims the use of natural iron-based minerals (lemonite, ferrihydrite, etc.) as durable catalysts or recycled iron based material (e.g. brake rotor) and sunlight energy as renewable heating source for a sustainable water treatment process. Chapter 6 suggests a common basis for future scientific research on Fe^0 remediation (as a whole). Chapter 7 lists cited references. The sixteen (16) peer-reviewed original research papers on which this thesis is formulated are included in appendix.

The experimental research was carried out at the Department of Chemistry of the American University of Beirut between February 2007 and February 2013 and was funded by the Office of Grants and Contracts at AUB (Grants number 788143, 888113, 988100, A88140, 188114, 288134), The Lebanese National Council for Scientific Research (Grants Number 522316, 522224) and the NAS-USAID (PEER Grant number 102548).

I would like to thank Dean McGreevy and my colleagues at the chemistry department for their support during this period.

A deep acknowledgment to Professor Jean-Claude Moutet, President of the Chemistry Habilitation Committee at Joseph Fourier University (Grenoble) for his help in according me the authorization to apply for the HDR diploma.

I am also very thankful to Dr. Alice Bejjani from the Lebanese Atomic Energy Commission for her fruitful collaboration.

My acknowledgements also go to my colleagues of the Department of Chemistry at the American University of Beirut and to the personnel of Kamal Al Shair Central Research Science Laboratory especially Dr. Youssef Mouneimne, Engineers Raja Chaaban and Joan Youness, Senior technicians Adnan Ruzz and Boutros Ghandour in addition to Rania Osta and Boutros Sawaya. They offered me really a peaceful atmosphere to accomplish this work.

I also would like to thank all my co-authors. Without their efforts, this work hasn't been accomplished. Most of them were undergraduate students and are currently pursuing graduate studies at AUB or abroad: Hala Abou Assi, Al Muthanna Tuqan, Habib Baydoun, Peter Dermesropian, Nadine Kibbi, Saly Geryes, Sara Bdeir, Ghada Ayoub and Sahar Naim. Data on metal hydroxide solubility were kindly provided by Professor Alison E. Lewis (University of Cape Town – South Africa).

Special thanks to Claude Casellas for encouraging me to rapidly apply for the Habilitation degree and to the reviewers for their valuable time in evaluating this work.

I also would like to thank Professor David Sedlak from the University of Calibronia – Berkeley for offering me the possibility of spending my Fulbritgh scholar award in 2009-2010 in his laboratory and for the actual collaboration under USAID-PEER program.

A very special thanks to Professor Chicgoua Noubactep from the University of Gottingen (Germany) for his insightful comments and his proof-reading on the present work.

Abstract

The permanent quest for affordable, applicable and efficient technologies for water treatment has culminated in the use of metallic elements as universal remediation materials. Contaminants of concern are biological (e.g. bacteria, viruses), chemical (organic and inorganic compounds) and physical (e.g. color, turbidity) in nature. The current classification of chemical contaminants is confusing as it accounts for (i) their origin/use (e.g. dyes, pesticides), (ii) the time scale at which their relevance was realized (e.g. emergent contaminants including pharmaceutical and personal care products) or simply the organic or inorganic nature. However, the physico-chemical nature of individual contaminants should be considered. Relevant intrinsic characteristics include: (i) charge, (ii) chemical reactivity, (iii) polarity, (iv) size and (v) structure. Intrinsic characteristics determine the interactions of each species with the environment. These interactions comprise the decontamination process. Relevant decontamination processes include adsorption, co-precipitation, oxidation, precipitation, reduction, and size-exclusion. From these processes, oxidation and reduction are not removal processes at µg/L-level. In other words, whenever oxidation and reduction are involved, contaminant removal must be guaranteed by one or a combination of the five other processes.

The universal character of metallic elements for water treatment arises from the fact that all interactions described above are likely to occur in a metal/water system. In particular, (i) native metal hydroxides are progressively generated and co-exist with primary iron corrosion products (e.g. Fe^{2+} and H_2/H), (ii) in the presence of native metal hydroxides, metal ions (e.g. Fe^{2+}) and adsorbed hydrogen are powerful reducing agents, and (iii) metal hydroxide precipitation and crystallization occurs in the presence of contaminants that are simply enmeshed, even though their adsorptive affinity to metal hydroxides might be weak. Clearly, the metal/water system is a dynamic system which huge potential for environmental remediation is yet to be exploited. The present work is focused on metallic iron (Fe^{0}) as a universal material showing successful use in water filtration systems and in advanced oxidation processes (AOPs), e.g. Fenton and persulfate. The elucidation of the chemical reactivity of the Fe^{0}/H_2O system is crucial for the understanding of the interactions of contaminants therein.

Based on the above assertion and the experimental evidence that Fe^0 and Fe^0 -based systems can be used as adsorbents generators and peroxide/persulfate activators to enhance the longevity of AOPs toward sustainable water treatment technologies, the objectives of the present work are: (i) to show why discrepancies exist among published data regarding Fe^0 technologies and its application to a wide range of contaminants; (ii) to demonstrate that Fe^0 and Fe^0 -based bimetallic (multimetallic) systems are operating in aqueous solutions on a basis of adsorption of contaminants after the formation of a porous nascent oxide scale of volumetric expansion/regression properties rather than pure reduction; (iii) to confirm the beneficial properties of Fe^0 particles, bimetallic and trimetallic Fe^0 -based systems of different sizes (nm, µm and mm) in order to activate hydrogen peroxide and persulfate into powerful radical oxidants for micro-contaminants degradation and (iv) to demonstrate that thermally and chemically activated persulfate is a successful technology that can be used for long term applications to overcome pollution of aquifers in organic contaminants toward more sustainable solutions.

Table of content

| Preface | 3 |
|---|---------|
| Abstract | 6 |
| 1. Overview on water remediation techniques | _10 |
| 1.1. Introduction | _10 |
| 1.1.1. Worldwide water situation: an overview | _10 |
| 1.2. Existing decontamination processes | _11 |
| 1.3. Fe ⁰ /H ₂ O as stand-alone remediation systems | _11 |
| 1.4. Advanced Oxidation Processes (AOPs) | _13 |
| 1.4.1. Fenton's reagent and Fenton like US assisted processes | _13 |
| 1.4.2. Persulfate oxidation through thermal and/or chemical activation | _14 |
| 2. Iron corrosion for environmental remediation | _15 |
| 2.0 Preamble | _15 |
| 2.1 State-of-the-art knowledge on Fe^0/H_2O systems | _16 |
| 2.1.1 Remediation Fe ⁰ /H ₂ O systems | _16 |
| 2.1.2 AOP Fe ⁰ /H ₂ O systems | _17 |
| 2.2 Aqueous iron corrosion | _18 |
| 2.2.0 Fe ⁰ materials for environmental remediation | _19 |
| 2.2.1 Electrochemical corrosion: Galvanic cells | _19 |
| 2.2.2 Interactions within a Fe ^{0} /H ₂ O system | _20 |
| 2.2.3 Contaminant removal in Fe ⁰ /H ₂ O systems | _22 |
| 2.3 Questioning the prevailing premise | _23 |
| 2.3.1 Fe ⁰ as reactive material | _23 |
| 2.3.2 Alternative materials | _24 |
| 2.4 Conclusions | _25 |
| 3. Overview on the appended articles on Fe ⁰ remediation | _26 |
| 3.1 Introduction | _26 |
| 3.2 Removal of PPCPs in $Fe^{0}/H_{2}O$ systems: The state-of-the-art | _27 |
| 3.3 Removal of PPCPs in Fe ^{0} /H ₂ O systems: Beyond the state-of-the-art | _29 |
| 3.3.1 Dancing in the round | _30 |
| 3.3.2 In tune with the mainstream science | _31 |
| 3.3.2.1 The Fe ^{0} /MnO ₂ /H ₂ O system | _31 |
| 3.3.2.2 Oxic vs. anoxic conditions | _31 |
| 3.3.2.3 Passivation of Fe ^o materials | _32 |
| 3.4 Significance for future research | _32 |
| 4. Nanoscale and microscale iron particles for the improvement of the Fenton process und | er |
| sonicated and silent systems | _34 |
| 4.1. Introduction | _34 |
| 4.2. Fundamentals of Fenton chemistry | 34 |
| 4.2.1. Mechanisms | _35 |
| 4.2.2. Types of Fenton's reaction | -37 |
| 4.2. Fundamentals of ultra-sonication chemistry | _38 |
| 4.3. Coupling sonochemistry with Fenton using Fe [°] (IFOP) | -39 |
| 4.4. Degradation of PAMs by IFOP | 40 |
| 4.4.1. Case of CBZ | 40 |
| 4.4.2. Case of SMX | _41 |
| 4.4.2.1. μ Fe ^v /H ₂ O ₂ systems (Silent vs US) | _42 |
| 4.4.2.2. nFe [°] /H ₂ O ₂ systems (Silent vs US) | _42 |
| 4.4.2.3. Identification of the transformation product | _42 |

| 4.5. Conclusion: nFe ⁰ vs μ Fe ⁰ in Fenton-like oxidation | 42 |
|--|--------|
| 5. Persulfate for the degradation of organic contaminants: Activation through therma | al and |
| chemical ways using zero-valent iron particles | 44 |
| 5.1. Introduction | 44 |
| 5.2. Fundamentals of Persulfate Chemistry | 45 |
| 5.2.1. Non activated persulfate | 45 |
| 5.2.2. Activated persulfate | 45 |
| 5.2.2.1. Thermal activation | 46 |
| 5.2.2.2. Chemical activation | 47 |
| 5.2.2.3. UV light and sun light activation | 48 |
| 5.2.2.4. Ultrasonic (US) PS activation | 48 |
| 5.2.2.5. Other PS activation methods | 49 |
| 5.2.2.5.1. Microwave (MW) | 49 |
| 5.2.2.5.2. Electron beam (EB) | 49 |
| 5.3. Application to PPCPs | 49 |
| 5.3.1. Case of Bisoprolol (TA-PS) | 50 |
| 5.3.2. Case of Ibuprofen (TA-PS) | 51 |
| 5.3.3. Case of Methylene Blue (TA-PS) | 52 |
| 5.3.4. Case of Sulfamethoxazole (CA-PS) | 53 |
| 5.4. Conclusions | 53 |
| 5.5. Future developments | 54 |
| 6. Concluding remarks | 56 |
| 7. Epilogue | 59 |
| 8. References | 60 |
| 9. Appended articles | 79 |

1. Overview on water remediation techniques

1.1. Introduction

1.1.1. Worldwide water situation: an overview

Save drinking water and sanitation are the primary source of human health, prosperity, and security. The provision of clean water to population is a global challenge for which scientific efforts should constantly support political decision making. The world's population is expected to reach about nine billion by around 2050. If the living standards continue to rise, the requirement of potable water for human consumption will amount to the resources of about three planet Earths (Dyson 1996, Daigger 2003, Hussam 2013). Water bodies are actually receiving from point and non-point sources per day an estimated amount of 2 million tons of anthropogenic waste including industrial waste (e.g. dyes, heavy metals, pharmaceuticals), human waste (e.g. faeces, urine) and agriculture waste (e.g. pesticides). Natural contamination is also an issue of concern. Relevant natural contaminants include arsenic, fluoride, pathogens (e.g. bacteria and viruses) and uranium (You et al. 2005, Shannon et al. 2008, Diao and Yao 2009, Momba et al. 2009, Litter et al. 2010, Giles et al. 2011, Noubactep 2011a, Shafiquzzaman et al. 2011, Ali 2012, Michen et al. 2012, Shi et al. 2012, Chiu 2013, Neumann et al. 2013). Accordingly, there is an increasing challenge to improve and preserve the quality of available water resources in order to prevent fresh water scarcity (Droste 1997, Franklin 1991).

In tune with scientific progress, including in chemical analytic, the focus of the water remediation industry constantly changes. For example, for some two decades, pesticides and other agricultural products were regarded as 'emergent contaminants'. However, their concern for water quality has decreased in recent years thanks to strict science-based regulations (Daughton 2002). Specific attention has been given to Pharmaceutical and Personal Care Products (PPCPs). The relevance of PPCPs as 'emergent contaminants' of worldwide concern relies on their worldwide spreading and their still unknown environmental fate. Moreover, because of their beneficial effects on human health, it is not likely that PPCPs will be successfully restricted in the same way as pesticides (Daughton 2007). Also, their consumption has an increasing trend with the increasing average age of the population (Daughton and Ternes 1999). The present work is mostly focused on the removal of PPCPs from water.

Residues of PPCPs were detected in a multiplicity of matrices using advanced analytical techniques (Petrovic et al. 2005, Hao et al. 2007, Klosterhaus et al. 2013) constituting

therefore a potential risk for aquatic and terrestrial organisms (Ternes 1998, Hirsch et al. 1999, Brown et al. 2006, Feitosa-Felizzola and Chiron 2009, Klosterhaus et al. 2013). Some of them, especially antibiotics, can also cause resistance in bacterial population, making them inefficient in the treatment of many diseases (Schwartz et al. 2003, Martinez 2009, Rosenblatt-Farrel 2009).

1.2. Existing decontamination processes

To prevent contamination of water matrices, researchers have developed different processes able to remove/degrade aqueous contaminants. Among those techniques one can enumerate conventional treatments including (i) biological processes via waste water treatment plants (WWTPs) including filtration and coagulation/flocculation/sedimentation (Adams et al. 2002, Arikan 2008), (ii) oxidation and advanced oxidation processes (AOPs) including chlorination (Sharma 2008, Acero et al. 2010), ozonation (Andreozzoi et al. 2005, Cokgor et al. 2004), Fenton (Andreozzi et al. 2010), ozonation (Andreozzoi et al. 2005, Cokgor et al. 2004, Gonzalez et al. 2007, Bautitz and Noguerira 2010), photolysis (Boreen et al. 2004, Jiao et al. 2008) and semiconductor photocatalysis (Calza et al. 2004, Klauson et al. 2010), electrochemical processes (Hiorse et al. 2005, Jara et al. 2007), adsorption (Crisafully et al. 2008, Putra et al. 2009, Kim et al. 2010), reverse osmosis (Li et al. 2004, Radjenovic et al. 2008), nano and ultrafiltration (Kosutic et al. 2007, Koyuncu et al. 2008), ion exchange (Choi et al. 2007, Ustun et al. 2007) and some combined processes (Klavarioti et al. 2009, Augugliaro et al. 2005).

Of the enumerated processes, the present work will focus on such involving the Fe^0/H_2O systems. Those are systems in which elemental iron (Fe^0) is used either as stand-alone remediation agent or as agent to enable or sustain remediation by other processes. A brief description of individual processes will be given below.

1.3. Fe⁰/H₂O as stand-alone remediation systems

Elemental iron (Fe⁰) has been introduced in the remediation as a reactive filler in subsurface permeable reactive barriers (PRBs). The initial goal was to reductively transform chemical species to less harmful, less mobile/soluble and/or more biodegradable species. For example, reduction of the chlorinated solvent trichloroethene (TCE) by Fe⁰ is reported as an abiotic process that mainly follows a beta-elimination mechanism with acetylene as an intermediate and ethene as the final product (Bigg and Judd 2000, ITRC 2011, Chen et al. 2012a, Chen et al. 2013). Accordingly, the reductive process, mostly randomly interchanged with the elimination process, is considered a reductive transformation resulting in

dechlorination/hydroxylation (Gillham and O'Hannesin, 1994, Matheson and Tratneyk, 1994). This chemical reaction has been reported to occur at the Fe⁰ surface and Fe⁰ considered the electron donor (direct reduction) (Weber 1996, Arnold and Roberts 1998, Arnold and Roberts 2000a, Arnold and Roberts 2000b).

The elimination of species without redox properties in $Fe^{0}/H_{2}O$ systems (Jia et al. 2007, Noubactep 2009a, Noubactep 2011b, Noubactep 2011c, Scott et al. 2011, Miyajima 2012, 2015) has challenged the view that contaminant Miyajima and Noubactep elimination/removal in PRBs is a reductive process. New paths toward the elucidation of the fundamental mechanisms of contaminant removal in Fe⁰/H₂O systems have been paved (Noubactep 2009b, 2010a, 2011b, 2011c, 2012, 2014a, 2014b, 2015). In this effort several branches of science involving aqueous iron corrosion were considered. These included electrocoagulation (Li et al. 2012, Moreno et al. 2009, Noubactep and Schöner 2010), corrosion science (Wilson 1923, Vernon 1945, Caule and Cohen 1959, Sato 2000, Sato 2001, Nesic 2007, Odziemkowski 2009, Schindelholz and Kelly 2012), synthetic organic chemistry (Werner 1951, Werner 1961, Noubactep 2012) and hydrometallurgy (Noubactep 2009c, Noubactep 2010c, Crane and Noubactep 2012). The process of iron corrosion and its volumetric expansive nature has been considered as well (Noubactep 2010c, Caré et al. 2013, Domga et al. 2015). In particular, the proper consideration of the expansion/compression cycles accompanying the well-documented volumetric expansion has enabled the description of Fe⁰ PRBs as filtration systems in which contaminant removal is improved by adsorptive size-exclusion (Miyajima 2012, Bilardi et al. 2013a, Bilardi et al. 2013b). The net result is that the mechanism of contaminant removal in $Fe^{0}/H_{2}O$ systems is fundamentally revisited. Relevant removal mechanisms included adsorption onto iron corrosion products, coprecipitation (enmeshment, entrapment or sequestration) of contaminants into precipitating corrosion products and adsorptive size exclusion (Noubactep 2007, 2008, 2009a, 2010a, 2010b, 2011a, 2011b, 2011c, 2012, 2013a, 2013b, 2013c, 2014a, 2014b, 2015).

The presentation in this section clearly shows that immersed reactive Fe^0 creates a dynamic cumulating in contaminant removal. The prevailing view is still the one that Fe^0 acts as a reducing agent for oxidized contaminants (Henderson and Demond 2007, Painter and Milke 2007, Tanboonchuy et al. 2012, Cai et al. 2013, Correia de Velosa and Pupo Nogueira 2013, Henderson and Demond 2013, Yu et al. 2013, Tepong-Tsindé et al. 2015a, Tepong-Tsindé et al. 2015b). This initial discrepancy on the nature of the processes yielding contaminant removal in Fe^0/H_2O systems hinders progress in research as it has a direct impact on all operational parameters that influence contaminant removal (Noubactep 2013c). Relevant

operational parameters include: (i) the intrinsic reactivity of Fe^0 , (ii) the availability of dissolved oxygen (e.g. oxic/anoxic systems), (iii) the availability of humic substances, inorganic species, manganese oxide, carbonates and phosphates.

The author has tested several PPCPs (e.g. Flutriafol, clofibric acid, diclofenac, dichlorophen, β -lactam antibiotics) to demonstrate the validity of the adsorption/coprecipitation model for the process of contaminant removal in Fe⁰/H₂O systems (Noubactep 2007, 2008). Tested systems include Fe⁰, bimetallic and trimetallic Fe⁰-based combinations. The availability of iron corrosion products were controlled in some key experiments by using controlled amounts of MnO₂.

1.4. Advanced Oxidation Processes (AOPs)

Advanced oxidation processes (AOPs) are common techniques used for the degradation of organic contaminants. AOPs are usually a pre-treatment in efforts of pollution abatement and rarely stand-alone remediation options. Several methods have been explored based on the provision of the medium with an oxidant that might work alone or with the help of a catalyst. The latter allows the production of powerful free radicals (e.g. Hydroxyl, Sulfate, Peroxyl, etc.) more or less selective with different rate constants that depend on the targeted contaminant (Andreozzi et al. 1999). Hydroxyl and sulphate radicals will be considered through this work upon investigation into (i) Fenton's reagent and Fenton-like reaction using Fe^0/H_2O_2 systems under sonicated and silent operational conditions and (ii) persulfate oxidation upon thermal and chemical activation. Pharmaceuticals (e.g. bisoprolol, ibuprofen, sulfamethoxazole) and dyes (e.g. Methylene blue) usually found in water ways and drinking water (Ternes 1998, Hernando et al. 2006) will be considered as selected probes for this investigation.

1.4.1. Fenton's reagent and Fenton like US assisted processes

Known by the iron-catalyzed hydrogen peroxide system, the Fenton's reagent is an experimental device releasing very strong but non-specific hydroxyl radicals oxidants into the aqueous phase. Those can be produced through two ways: (i) by Fenton oxidation using soluble iron such as dissolved ferrous iron (Fe^{2+}) or (ii) by Fenton-like process using iron oxyhydroxide such as goethite or even magnetite (Fe_3O_4) magnetic nanoparticles (MNPs) (Dhakshinamoorthy et al. 2012). The iron-based catalyst is here playing a key role in activating hydrogen peroxide into hydroxyl radicals. Such activation can be done homogeneously or heterogeneously. The former is based on dissolved Fe^{2+} while the latter of greater importance can be based on the use of iron particles (Fe^0) or iron-derived particles of

micrometric and/or nanometric scale allowing easier separation from the effluent and, eventually, reuse (Corma and Garcia 2003). Optimization of the oxidation process should take into account agglomeration of the heterogeneous catalyst so as low frequency US (40 kHz) is used to assume (i) homogeneous distribution of the solid catalyst into the reactive medium and (ii) non accumulation of an oxide scale inhibiting therefore corrosion products to be released into the bulk solution such as Fe^{2+} . Additional factors including pH, ionic strength and ions identity are also of main importance for the optimization of the heterogeneous catalytic oxidation. Carbamazepine has been used as pharmaceutical probe as well as sulfamethoxazole.

1.4.2. Persulfate oxidation through thermal and/or chemical activation

Recently introduced for application to effluents and soils contaminated with organic compounds, persulfate (PS) use becomes of universal interest (Huang et al. 2005). PS can be used to degrade organic contaminants however after being activated into sulfate radicals (Liang et al. 2004). Those are powerful oxidants more selective than hydroxyl radicals and work by electron abstraction instead of hydrogen abstraction as it is the case of hydroxyl radicals (Minisci and Citterio 1983). Activation can take place thermally by simple heat of the solution (Liang et al. 2003), photo-chemically through UV irradiation (Lin et al. 2011a) or chemically after adding appropriate amount of a specific transition metal such as cobalt, silver or iron (Anipsitakis and Dionysiou 2004). Being very selective, sulfate radicals can even degrade the transformation products toward full mineralization that is the conversion of the original compounds into CO2 and H2O. However, this AOP has also some limitations especially related to the experimental conditions in terms of pH, ionic strength, radical quenching related to an excess of catalytic cations (Fe^{2+} , Co^{2+} , Ag^+). Accordingly, optimization of this process will be presented through the selection of different pharmaceutical probes (Bisoprolol, Ibuprofen, Sulfamethoxazole) and a universal dye (Methylene Blue) in addition to iron-based catalysts including monometallic (Fe⁰), bimetallic (Fe⁰Ag⁰, Fe⁰Co⁰) and trimetallic systems (Fe⁰Ag⁰Co⁰, Fe⁰Co⁰Ag⁰). These systems are presented as an alternative to the use of dissolved catalysts where PS activation is extremely rapid resulting in cannibalization of radicals and very low reaction stoichiometric efficiency (RSE) defined as the number of molecules of probe degraded over the number of molecule of PS consumed (Δ [Probe]/ Δ [PS]).

2. Iron corrosion for environmental remediation

2.0 Preamble

A survey of the sixteen papers on which this work is based reveals that a very recent article (P12 – Caré et al. 2013) deals with fundamental theoretical considerations of processes of iron corrosion. It may be surprising that for a 30-years-old technology, basic aspects of involved processes are still discussed. The situation is even worse as P12 (Caré et al. 2013) was written as collective response of five of the six authors to numerous colleagues acting as peer-reviewers at several high impact scientific journals to which articles post P3 (Ghauch 2008) were submitted. The expertise of the sixth author (Dr. Caré) is "iron corrosion in steel rebar" with particular emphasis on concrete cracking caused by expansive iron corrosion (Caré et al. 2008, 2010).

The views presented in P12 (Caré et al. 2013) are controversially discussed in the literature although they are based on the first principle of chemical thermodynamics (iron mass balance). The still prevailing view, that metallic iron is an environmental reducing agent, is based on the thermodynamic valid argument, that the redox potential of the electrode reaction Fe^{II}/Fe^0 ($E^0 = -0.44$ V) is lower than that of dissolved oxygen ($E^0 = 0.81$ V) and several oxidized species available as contaminants. However, water (H_2O or H^+) is also an oxidant for Fe^0 ($E^0 = 0.00$ V). Because water is the solvent, its stoichiometric abundance makes Fe^0 oxidative dissolution by H_2O the most favourable reaction in the Fe^0/H_2O system. The consequence is that the Fe^0 surface is constantly covered by iron oxides and Fe^0 is not (readily) available for electron transfer to the contaminant (direct reduction, electrons from the metal body).

It is important to notice that the view that Fe^0 is an environmental reducing agent has: neither properly considered (i) the thermodynamics of iron oxide precipitation nor (ii) the thermodynamics of processes occurring within the oxide scale at the vicinity of Fe^0 . Although these aspects have been presented in the scientific literature six years ago (Noubactep 2007, 2008), there is evidence that this alternative view is not really understood by the majority of active researchers on "Fe⁰ for environmental remediation". Therefore, the present chapter aims at reviewing the essential features of aqueous iron corrosion and outlining their significance for processes yielding contaminant removal or contaminant oxidation in Fe^0/H_2O systems. Moreover, the key features will be contrasted by the state-of-the-art knowledge on the operating mode of Fe^0/H_2O systems.

2.1 State-of-the-art knowledge on Fe⁰/H₂O systems

2.1.1 Remediation Fe⁰/H₂O systems

Permeable reactive barriers (PRBs) containing granular metallic iron (Fe⁰) are regarded as an accepted technology for the remediation of contaminated groundwater. Fe⁰ PRBs operate with natural groundwater gradients driving flow to treat large volumes of water with low concentrations of contaminants (Bigg and Jugg 2000, O'Hannesin and Gillham 1998, Scherer et al. 2000, Jambor et al. 2005, Painter and Milke 2007, Henderson and Demond 2013). The most commonly used Fe⁰ is of mm or μ m size and represents a standard material (conventional Fe⁰) against which alternatives are compared. Alternative materials included nano-sized Fe⁰ (nano-Fe⁰) and multi-metallic systems (including nano-sized). Other reactive metals (e.g. Al⁰, Cu⁰, Mg⁰, Mn⁰, Ni⁰ and Zn⁰) have been also tested at all particle sizes (nm, μ m and mm) (Bokare et al. 2013, Du et al. 2013, O'Carroll et al. 2013). A general trend is that nano-Fe⁰ are more efficient than Fe⁰, bimetallic more efficient than Fe⁰, and Zn⁰ is more efficient than Fe⁰ (Grittini et al. 1995, Liang et al. 1997, Elliot and Zhang 2001, Lien and Zhang 2002, Begum and Gautam 2011, Salter-Blanc and Tratnyek 2011, Bokare et al. 2013, Cai et al. 2013).

 Fe^{0} is considered a (strong) reducing agent which induces the reductive transformation of organic species or promote the reductive precipitation of oxidized metals (Henderson and Demond 2013). It is acknowledged that water and other species like nitrate or oxygen will quantitatively oxidize Fe^{0} . The net result may be the production of gases (H₂, N₂). In addition, the pH increase inherent to iron corrosion has been reported to result in the precipitation of non-contaminant phases (e.g. CaCO₃, FeCO₃). Because species like calcium and carbonate are typically present at concentrations several orders of magnitude higher than that of target contaminants, it is commonly reported that non-contaminant phases dominate the precipitates that are formed in field $Fe^{0}/H_{2}O$ systems (Westerhoff and James 2003, Huang and Zhang 2005, Henderson and Demond 2013).

The drawbacks of Fe^{0} PRBs have been identified. They include (i) lack of opportunities for external reactivity manipulation (**drawback 1**), as the efficiency of the system is mostly determined by the reactivity of used Fe^{0} , (ii) progressive Fe^{0} passivation by precipitation of minerals at the surface (reactivity loss) (**drawback 2**), and (iii) gradual decrease of the porosity and permeability of the system (permeability loss) (**drawback 3**) yielding at the term to system clogging (Morrison et al. 2006, Noubactep et al. 2010, Caré et al. 2013). To

properly address these drawbacks (drawback 1 through 3), the fundamental understanding of the mechanisms of the dynamic processes occurring in Fe^0/H_2O system must be realized.

Summarized, Fe^{0} is considered a reducing agent for the reductive transformation of oxidized contaminants. Fe^{0} is also oxidized by water and elevated pH values induce the precipitation of non-contaminant phases and the production of gases (H₂, N₂). At the term, non-contaminant phases and produced gases (H₂, N₂) significantly reduce the porosity, and thus, the permeability of Fe^{0} PRBs. It is essential to notice that this prevailing approach has never been univocally accepted (Lipczynska-Kochany et al. 1994, Bonin et al. 1998, Lavine et al. 2001, Furukawa et al. 2002, Jiao et al. 2009, Odziemkowski 2009, Liu et al. 2013). For example, using differential pulse polarography to simultaneously monitor the disappearance of nitrobenzene and the appearance of Fe^{II} , Lavine et al. (2001) could not clarify mechanistic details of processes yielding to the chemical reduction of "organic compounds at the Fe^{0} surface". Moreover, the reductive transformation model cannot explain why inorganic and organic species without redox properties are quantitatively removed in Fe^{0}/H_2O systems (Noubactep 2007, 2008). On the other hand, the whole discussion on the process of permeability loss has not properly considered the iron mass balance (Caré et al. 2013).

2.1.2 AOP Fe⁰/H₂O systems

Metallic iron (Fe⁰) has been shown an effective material for the improvement of advanced oxidation processes (AOPs). The idea behind using Fe⁰ for oxidant generation is that Fe^{II}, Fe^{III} and Fe^{II}/Fe^{III}-based minerals are known for their ability to serve as catalyst for the oxidative degradation of recalcitrant contaminants (Lipczynska-Kochany et al. 1995, Pham et al. 2009). Fe⁰ can induce a prolong activation of specific reactants like hydrogen peroxide (HP) or persulfate (PS) into powerful oxidative radicals e.g. Hydroxyl Radicals (HRs) and Sulfate Radicals (SRs).

Fenton (Segura et al. 2013, Cao et al. 2013), Fenton-like assisted sonication (Ziylan et al. 2013, Lai et al. 2013) and Persulfate (Hussein et al. 2012, Oh and Shin 2013) technologies are AOPs primarily using catalytic materials (e.g. Fe^0 - Eq. 1) to produce HRs (Eq. 2) and SRs (Eq. 3) from hydrogen peroxide and persulfate, respectively.

$$Fe^{0} + 2H^{+} +))) \rightarrow Fe^{2+} + H_{2}$$
 (in acidicsolution) (1)

$$H_2O_2 + Fe^{2+} \rightarrow \bullet OH + OH^- + Fe^{3+}$$
(2)

$$S_2O_8^{2-} + Fe^{2+} \rightarrow SO_4^{2-} + SO_4^{\bullet-} + Fe^{3+}$$
 (3)

 Fe^{0} aqueous oxidative dissolution (Eq. 1) in-situ generates iron corrosion products with welldocumented catalytic properties. Relevant catalytic species include Fe^{2+} , Fe^{3+} , iron oxides, iron hydroxides, iron oxo-hydroxides.

Most of these iron corrosion products are reactive (Sikora and McDonald 2000), allowing additional useful catalytic properties that are yet to be characterized under well-controlled experimental/operational conditions. It is essential to notice that Fe^0 is irreversibly consumed in AOP Fe^0/H_2O systems. However, Fe^0 consumption produce useful species (Fe^{2+} , Fe^{3+} , Fe oxides and hydroxides) that might be regenerated upon redox reactions. Furthermore, Fe^0 mass loading can be optimized to avoid that the maximum contaminant level (MCL) for dissolved iron (0.3 mg.L⁻¹) is not reached. This precaution avoids sludge generation through iron oxide precipitation.

Summarized, Fe^{0} can induce an optimal promotion of the degradation of recalcitrant organic contaminants. Fe^{0} improves the efficiency and longevity of AOPs once coupled to HP or PS in aqueous media. The key feature of using Fe^{0} in AOPs is the chemical reactivity of Fe^{0} that should be sustained to enable continue and progressive release of Fe-based activators of HP and PS.

2.2 Aqueous iron corrosion

Wealth of information is available on aqueous iron corrosion under natural conditions. They can be found in academic thesis, encyclopaedias, peer-reviewed scientific literature, technical reports and textbooks. Information on aqueous iron corrosion originates from various research branches including pipe corrosion in the oil and gas industry, waste repository industry and water supply industry (Snoeyink and Jenkins 1980, van Orden 1989, Brondel et al. 1994, Stratmann and Müller 1994, Antonio et al. 2000, Sikora and Macdonald 2000, Sarin et al. 2001, Sarin et al. 2004a, Sarin et al. 2004b, Nalli 2010, Obanijesu et al. 2010). However, profound understanding of the corrosion mechanism under remediation conditions is very important for material selection and design of sustainable systems (Gheju and Balcu 2011, Miyajima 2012, Bilardi et al. 2013a, Bilardi et al. 2013b, Btatkeu et al. 2013). In particular, the specificities of Fe⁰ as used in environmental remediation have to be considered in its totality before a decision is made on the proper materials (Reardon 1995, Noubactep et al. 2005a, Reardon 2005, Noubactep et al. 2009, Btatkeu et al. 2013). The present section is a consolidated information pooled from various sources for the benefit of properly designing 'Fe⁰ remediation' systems.

2.2.0 Fe⁰ materials for environmental remediation

The Fe^{0} remediation industry would like to design $Fe^{0}/H_{2}O$ systems able at uninterrupted treat groundwater for some decades (up to four). During this long service life, used Fe⁰ materials would undergo corrosion under varying environmental conditions due to seasonally changes in water compositions and temperatures. The key feature of Fe⁰ materials for environmental remediation is that mechanical properties (e.g. ductility, elongation, hardness, impact strength, proof stress, tensile strength) are not of any important significance (Feature 1). Only the chemical reactivity is important. This key feature suggests that purposeful materials for environmental remediation should be characterized for their intrinsic chemical reactivity (Btatkeu et al. 2013). The second important feature of remediation Fe^{0} is their availability as chips, fillings and granules whose thickness (< 4 mm) is hardly larger than the wall thickness of industrial iron pipes (Feature 2). Feature 1 suggests that available Fe⁰ materials are not optimised for environmental remediation as care was taken for optimal mechanical properties during their manufacture. On the other hand, Feature 2 suggests remediation Fe⁰ particles are considerably most reactive than Fe⁰ pipes because of their smaller particle size (larger surface area). However, a smaller particle size also means shorter distance for electrons transfer and higher passivation probability in comparison to (thicker and longer) Fe⁰ pipes for which anodic and cathodic regions for a galvanic cell may be some km distant (Snoeyink and Jenkins 1980).

Taken together, **Feature 1 and 2** clearly demonstrate that despite the abundance of literature on aqueous iron corrosion, remediation $Fe^{0}/H_{2}O$ systems is a stand-alone research area which can only intelligently use impetius from the broad corrosion knowlegde. It has already been discussed, that this task was not properly done (Noubactep 2007, Ghauch et al. 2011a, Gheju and Balcu 2011, Togue-Kamga et al. 2012, Caré et al. 2013, Noubactep 2013c). Hence it is imperative for Fe^{0} remediation research community to be familiar to basic aspects of iron corrosion concerning the systems they are striving to design (Noubactep 2013c). Such a generalized "Corrosion Awareness" would ensure a scientific basis for sustainable $Fe^{0}/H_{2}O$ systems. The chapter is an attempt in this direction.

2.2.1 Electrochemical corrosion: Galvanic cells

Electrochemical corrosion of Fe^0 materials involves a flow of electrons between cathodic and anodic areas on a piece of Fe^0 metal or several pieces of Fe^0 joined together by an electronic conductor (e.g. metal, metal oxide, activated carbon). A galvanic cell is created when two different metals (areas of the same alloys) develop a potential difference between them in a conducting electrolyte. The metal with the lower positive electrochemical potential (E^0 value) acts as an anode and corrodes metal ions away to balance the electron flow. The second metal with higher positive E^0 value acts as a cathode and is protected from corrosion.

The kinetics (and probably the extent) of corrosion in a galvanic cell depends primarily upon the difference in potentials of involved alloys/metals, their surface areas and the solution chemistry (conductivity, pH, temperature). The difference in potentials created between two metals depends on their relative ranking in the electrochemical series. Whenever two metallic materials are in contact and are exposed to an ionic conductive environment, the metal with most negative potential becomes anode and starts corroding. In other words, when Fe^0 ($E^0 = -$ 0.44 V) is alloyed with other metals with more positive potential ($E^0 > -0.44$ V), iron corrodes. Relevant metals include: Ag⁰, Cd⁰, Co⁰, Cu⁰, Ni⁰, Sn⁰, Pd⁰, Pt⁰. In other words, alloying or plating Fe^0 with one or more elements with $E^0 > -0.44$ V solely accelerates the kinetics of Fe⁰ corrosion. This evidence demonstrates that bimetallic and multi-metallic systems used in environmental remediation accelerate Fe⁰ corrosion by increasing the number of galvanic cells. Clearly, any increased chemical process (e.g. reduction) in a multi-metallic system relative to a mono-metallic system is only indirectly coupled to iron corrosion (statement 1). Statement 1 suggests that although accelerated iron corrosion (anodic reaction) of a multi-metallic system is an electrochemical process, observed contaminant reduction is not a simultaneous reaction (the corresponding cathodic reaction). Statement 1 also suggests/confirms that observed contaminant reduction by conventional Fe⁰ is not necessarily the cathodic reaction corresponding to anodic iron dissolution (Noubactep 2009d).

2.2.2 Interactions within a Fe⁰/H₂O system

This section describes Fe^0 corrosion under anoxic conditions and discusses the significance of iron corrosion for environmental remediation. Fe^0 oxidation by water is described by Eq. 4:

$$Fe^{0} + 2 H^{+} \Leftrightarrow Fe^{2+} + H_{2}$$

$$Fe^{0} + 2 H_{2}O \Leftrightarrow Fe^{2+} + H_{2} + 2 OH^{-}$$
(4)
(4)
(4)

A key feature of Eq. 4 is that both Fe^{2+} and H_2 are reducing agents. In other words if a species is chemically reduced in a Fe^0/H_2O system, it is difficult to segregate whether involved electrons are from Fe^0 , Fe^{2+} or H_2 . This fact was already acknowledged by Matheson and Tratnyek (1994) but further investigations have culminated in direct reduction (electrons from Fe^0) being favoured as major mechanism of contaminant reduction in Fe^0/H_2O systems (Ghauch et al. 2011, Gheju and Balcu 2011, Noubactep 2011b, Noubactep 2011c, Caré et al. 2013, Noubactep 2013a, Noubactep 2013b, Noubactep 2011d). In Eq. 4 water (H₂O or 2 H⁺) is the cathodic reaction coupled to anodic Fe^0 dissolution.

After Le Chatelier principle, Eq. 4 is shifted to the right (accelerated Fe^0 dissolution) if: (i) H^+ is abundantly available (acidic conditions), (ii) Fe^0 is more reactive (e.g. bimetallic and trimetallic systems – Section 2.1) or (iii) Fe^{2+} and H_2 are consumed. Fe^{2+} can be consumed by relevant oxidizing species ($E^0 > 0.77$ V) and the system will be enriched in Fe^{III} species including oxides and hydroxides. Fe^{II} and Fe^{II}/Fe^{III} oxides and hydroxides are porous precipitates with documented adsorptive affinity to both Fe^{2+} and H/H_2 (Charlet et al. 1998, Jiao et al. 2009). In particular, adsorbed Fe^{II} (or structural Fe^{II}) is a more powerful reducing agent than dissolved Fe^{II} (White and Paterson 1996, Charlet et al. 1998). The fact that reducing species are abundantly produced in Fe⁰/H₂O systems complicates mechanistic discussion and severely questions the prevailing view that Fe⁰ reduces organic species (reductive degradation) and sequestrates inorganic species (reductive precipitation) (Lackovic et al. 2000, Lavine et al. 2001). This assertion alone makes clear that it is yet not clear how organic species are removed as their quantitative removal was largely documented, even for species that are not reducible (Lai et al. 2006, Jia et al. 2007, Miyajima 2012). On the other hand, adsorption and co-precipitation of organics species by iron oxide is well-documented (Gu et al. 1994, Bojic et al. 2009, Noubactep and Schöner 2010, Eusterhues et al. 2011).

The presentation above suggests that chemical reactions responsible for contaminant reduction are the following (RX stands for a halogenated organic compound):

 $Fe^{0} + RX + H^{+} \Longrightarrow Fe^{2+} + RH + X^{-}$ (5)

$$2 \operatorname{Fe}^{2+}_{ads} + \mathrm{RX} + \mathrm{H}^{+} \Longrightarrow 2 \operatorname{Fe}^{3+} + \mathrm{RH} + \mathrm{X}^{-}$$
(6)

$$H_{ads} + RX \Longrightarrow RH + X \tag{7}$$

While Eq. 5 is still conventionally used to rationalize contaminant reduction in $Fe^{0}/H_{2}O$ systems (Liu et al. 2013), the occurrence of reactions according to Eq. 6 and 7 have been traceably demonstrated (Gould 1982, Charlet 1998, Jiao et al. 2009). Moreover, the occurrence of individual reactions depends on the accessibility of the reaction site. Fe^{0} is constantly covered by an oxide scale such that a contaminant (RX) must migrate through the oxide scale to reach the reactive site on Fe^{0} (Noubactep 2007, 2008). It is not likely, that contaminants will quantitatively migrate to the Fe^{0} surface. This assertion is supported by the results of Strathmann and Müller (1994) who demonstrated that even molecular oxygen cannot diffuse to the Fe^{0} surface. Thus, Fe^{0} is oxidized by water (Eq. 4) and oxygen is reduced by Fe^{II} (Eq. 6-like). In other words, even if the oxide scale is electronic conductive

(e.g. Fe₃O₄), the chemical reduction by Fe^{II} species is kinetically more favourable than the electrochemical reduction by Fe^{0} (statement 1 – section 2.1).

2.2.3 Contaminant removal in Fe⁰/H₂O systems

The presentation above has recalled that Fe^0 is mainly oxidized by the solvent (H₂O). Thus iron corrosion products are abundant in the system. Any contaminant entering in such an 'ocean' of iron oxides and hydroxides (Noubactep 2009a) could be adsorbed by already precipitated species or enmeshed in the matrix of precipitating hydroxides (co-precipitation). Thus, adsorption and co-precipitation are the fundamental mechanism of contaminant removal in Fe^0/H_2O systems (Noubactep 2010a). In a column system, size-exclusion is the third removal mechanism. Adsorptive size exclusion, coupled to adsorption and coprecipitation explain why all classes of contaminants could be quantitatively removed in Fe^0/H_2O systems. Recent experimental works have confirmed theoretical predictions, that the primary selectivity criterion is the contaminant affinity to iron corrosion products (Gheju and Balcu 2011, Scott et al. 2011, Miyajima 2012, Miyajima and Noubactep 2012, Bilardi et al. 2013a, Bilardi et al. 2013b, Btatkeu et al. 2013, Miyajima and Noubactep 2013). Clearly, Fe^0 based filtration systems are more suitable for contaminants with high affinity to iron hydroxides/oxides. This conclusion does not consider the chemical reducing power of Fe^0 and confirm the view that Fe^0 is a generator of removing agents.

Regarding Fe^0 as a generator of removing agents has rationalized the operating mode of remediation Fe^0/H_2O systems. It explains why all classes of contaminants are removed and intuitively dictates that more material amounts are used for species with less affinity to iron oxides. Moreover, it eases the design of filtration systems, as main and side reactions are no more significant. The major task is to sustain iron corrosion and by ricochet contaminant removal. Now how to sustain iron corrosion?

While focussing the attention of contaminant reduction, researchers have forgotten to chemically induce reductive dissolution of iron corrosion products to complete mass balance (Kishimoto et al. 2011). On the other hand, iron balance has not considered the volumetric expansive nature of iron corrosion (Pilling and Bedworth 1923). The volume of each corrosion product is 2.1 to 6.4 times larger than the volume of Fe^0 in the metal body (Caré et al. 2008). This volumetric expansion is the major cause of permeability loss (Bilardi et al. 2013a, Bilardi et al. 2013b, Caré et al. 2013) and not the accumulation of non-contaminant phases (CaCO₃, FeCO₃) and/or the accumulation of gas (H₂, N₂) (Henderson and Demond 2011).

The proper consideration of the volumetric expansive nature of iron corrosion has opened an avenue for sustainable Fe⁰-based filtration systems. It has confirmed that a 100 % Fe⁰ system may be efficient but not sustainable (Hussam and Munir 2007, Hussam 2009). Moreover, the reproducible observation of Hussam and Munir (2007) is rationalized. In a pure Fe⁰ system, all particles are volumetric expansive and no place "is left" for expansive iron corrosion. Calculations showed that such a 100 % Fe⁰ system is clogged before one half of Fe⁰ is consumed. Accordingly, besides being a non-sustainable system, a 100 % Fe⁰ system is a pure material wastage. These theoretical considerations have demonstrated unambiguously that admixing Fe⁰ to non-expansive (inert or reactive) materials is a prerequisite for sustainable filtration systems (Caré et al. 2013). First experimental results have shown that the optimal Fe⁰:additive volumetric ratio is 25:75 (Bilardi et al. 2013a, Miyajima and Noubactep 2013). This low Fe⁰ ratio makes the resulting system a Fe⁰-amended system in which Fe⁰ is the additive (e.g. Fe⁰-amended sand filter).

2.3 Questioning the prevailing premise

2.3.1 Fe⁰ as reactive material

The presentation above has severely questioned the validity of the prevailing paradigm that Fe^{0} is an environmental reducing agent. Moreover, the prevailing paradigm should be abandoned to booster advance in research and thus, acceptance of the Fe^{0} remediation technology (Noubactep 2011b, Noubactep 2011c, Noubactep 2011d, Noubactep 2013c, 2014a, 2014b). The major curse of the prevailing paradigm is that it has created and maintained confusion on the true mechanism of "contaminant reduction" and "contaminant removal" in $Fe^{0}/H_{2}O$ systems (Noubactep 2015). Acknowledging that Fe^{0} is a generator of removing agents has re-directed efforts to overcome or at least limit the identified drawbacks of the Fe^{0} remediation technology (drawback 1 through 3 – Section 2.1.1).

It is certain that external reactivity manipulation (**drawback 1**) is not possible but materials should be selected to be long-term reactive in the subsurface. Tasks to this end include plating, admixing Fe^0 with reactive oxides or granular activated carbon (Noubactep et al. 2012a). The reactivity loss (**drawback 2**) is unavoidable but material selection and system design should occur such that the residual Fe^0 reactivity and the thickness of the system will guarantee long term efficiency. Finally, permeability loss (**drawback 3**) is solved by saving Fe^0 material costs as the volume of Fe^0 corresponding to a pure Fe^0 system should be reduced at least to one half. However, admixing Fe^0 with non-expansive materials suggests that thicker filters should be built. This will be coupled to higher installation costs. Accordingly, even the installation costs of permeable reactive barriers could be revised.

2.3.2 Alternative materials

The search for alternatives to Fe^0 was based on the premise that Fe^0 ($E^0 = -0.44$ V) is an environmental reducing agent. Accordingly, mostly tested alternatives are two metallic elements with more negative E^0 values: AI^0 ($E^0 = -1.66$ V) and Zn^0 ($E^0 = -0.76$ V). However, these efforts have ignored the hart fact that iron hydroxides are the sole low soluble hydroxides in the whole pH range of natural waters (Lewis and Swartbooi 2006, Lewis and van Hille 2006, Lewis 2010, Noubactep 2013a, Noubactep 2013b) (Fig. 1). Figure 1 suggests that $Fe(OH)_2$ is more soluble than $Zn(OH)_2$, however, because of the very strong hydrolysis and polymerisation of Fe^{2+} species, the concentration of dissolved Fe^{II} is always comparable to that of Fe^{III} . In other words, testing AI^0 and Zn^0 has not properly considered the thermodynamics of involved species. On the other hand, it has been traceably demonstrated that, despite more positive E^0 values, iron corrodes under environmental conditions because oxide scales on Fe^0 are neither adherent nor impermeable (Dickerson et al. 1979, Sato 2001, Noubactep 2010b). The fact that AI^0 and Zn^0 are likely to be passivated in the middle term questions the significance of results from short term batch and column laboratory experiments for field situations.



Figure 1. pH dependence of metal hydroxide solubility. The presentation is limited to the range of natural waters ($4.5 \le pH \le 9.5$). A clear pH dependence of hydroxide solubility can be observed. Considered minerals are Al(OH)₃, Fe(OH)₂, Fe(OH)₃ and Zn(OH)₂. (Data from Lewis 2010).

2.4 Conclusions

This chapter has clearly shown that the description of processes leading to contaminant removal in $Fe^{0}/H_{2}O$ systems are currently very empirically based. There is an urgent need to improve the scientific understanding of these processes for control, optimisation, scale-up and at term proper design of efficient and sustainable Fe^{0} remediation systems. The coming chapters will summarize the contribution of the author to the still widely ignored state-of-the-art knowledge on the processes governing contaminant removal in $Fe^{0}/H_{2}O$ systems (Comba et al. 2011, Gheju and Balcu 2011, Scott et al. 2011, Caré et al. 2013, Noubactep 2013a, 2013, 2013c). It is clearly demonstrated that the suitability of Fe^{0} for contaminant removal and oxidation process results from the long term production of reactive species. Thus, in both cases, Fe^{0} is an in place (in-situ) generator of remediation agents.

3. Overview on the appended articles on Fe⁰ remediation

3.1 Introduction

2007 was a crucial year for the development of the Fe⁰ remediation technology. Two complementary peer-reviewed critical review articles were published during this year. The first article by Henderson and Demond was published in the established journal "Environmental Engineering Science" (Vol. 24). The second article by Noubactep was published in the first issue of "The Open Environmental Journal".

Contrarily to previous overview articles/reports analysing individual case studies (e.g. Scherer et al. 2000, Richardson and Nicklow 2002, Tratnyek et al. 2003, Warner and Sorel 2003), Henderson and Demond (2007) rather considered broad performance issues. Accordingly, they analysed available data from field Fe^0 PRBs and determined parameters contributing to system failure. Their analysis supported the view that Fe^0 reactivity loss, rather than a decrease of the system's hydraulic conductivity (permeability loss), is more likely to limit PRBs sustainability (long-term efficiency). They concluded that, because of the sparseness of field monitoring of relevant parameters (including Eh), their results needed to be corroborated by future data. It is essential to state that the review of Henderson and Demond (2007) (105 independent citations at Scopus on 2013/07/01) was based on the unrestricted view that Fe^0 is a reducing agent. Moreover, as shown in Chap 2, the volumetric expansive nature of iron corrosion was not properly considered in discussing permeability loss.

Noubactep (2007) (Scopus not indexed) came along with a different view stating that the well-documented contaminant reductive transformation results from indirect reduction by primary iron corrosion products (e.g. Fe^{II} and H/H_2). Accordingly, iron is corroded by water and contaminant reductive transformation occurs within the oxide scale on Fe^0 . This oxide scale is regarded by the reductive transformation model as a curse (reactivity loss). The model introduced by Noubactep (2007, 2008) belittled the importance of contaminant reductive transformation and revealed the essential role of adsorption and co-precipitation in the process of contaminant removal. Adsorption and co-precipitation were mostly considered as side processes in a reductive system (Mantha et al. 2001, Furukawa et al. 2002). It is essential to notice that Noubactep used a system-based approach (which is contrary to the still common approach) to determine the suitability of Fe^0 for individual contaminants in treatability tests (Richardson and Nicklow 2002, Oh 2009, Huang et al. 2013a).

The present work is a contribution to the global effort to optimise the efficiency of Fe^{0} remediation systems (including Fe^{0} PRBs). This chapter will summarize the author's

achievements, which are presented in details in the appended original articles P1 through P10. Results were all obtained in batch experiments but their contribution to properly design Fe^{0} filtration systems (Chap. 2) is obvious.

The presentation will follow a chronological perspective to clarify how the author has liberated himself from the still prevailing model that Fe^0 is an environmental reducing agent to one of the main co-developers of the alternative model regarding Fe^0 as generator of removing (and reducing) agents. The state-of-the-art knowledge on the treatment of PPCPs will first be presented.

3.2 Removal of PPCPs in Fe⁰/H₂O systems: The state-of-the-art

During the past 14 years, an increasing awareness of the health impacts of pharmacological pollution has been observed (Daughton and Ternes 1999, Ternes and Hirsch 2000, Heberer 2002, Ternes et al. 2004, Westerhoff et al. 2005, Pauwels and Verstraete 2006, Stieber et al. 2008, Kümmerer 2009, Stieber et al. 2011, Kaplan 2013, Zhang et al. 2013). Conventional (waste)water treatment technologies are not able to efficiently address pharmacological pollution, which is collectively termed pharmaceuticals and personal care products ("PPCPs") (Daughton and Ternes 1999). A specific treatment of hospital wastewater should minimise the emission of PPCPs. Various strategies have been introduced to mitigate the extent of contamination due to PPCPs. One approach is to incorporate a Fe⁰ remediation unit.

The idea of using Fe^0 for PPCPs mitigation relies on the premise, that PPCPs chemical reduction ideally leads to non-bioactive and readily biodegradable transformation products, which can then be treated in conventional wastewater treatment plants. The same idea has been followed to demonstrate the suitability of Fe^0 for individual biological and chemical contaminants (Richardson and Nicklow 2002). However, given the huge number of PPCPs (more than 30,000) that are additionally present as mixtures in hospital wastewaters, it is questionable whether all PPCPs could be individually tested according to the present approach (Hendriks 2013). Admittedly, some published works have tested the suitability of Fe^0 for several PPCPs. For example, Stieber et al. (2011) tested the suitability of a granular commercial Fe^0 material (Gotthard Mayer Metallpulver GmbH, Rheinfelden/Germany) for the treatment of 7 PPCPs. However, the selection of the tested PPCPs was not based on their affinity to iron corrosion products. In fact, to be quantitatively transformed in a Fe^0/H_2O system, any reducible contaminant must not be repulsed by iron oxides (covering the Fe^0 surface). In other words, the adsorptive affinity to iron oxides should be the paramount

criterium to segragate species treatable in Fe^0/H_2O systems (Scott et al. 2011, Miyajima 2012).

The evidence that hydrophilic and poorly degradable organics survive wastewater treatment corroborates the importance of electrostatic interactions in the process of contaminant removal. This evidence has been known for decades (Mitchell et al. 1955) although its proper consideration in several aspects of environmental science is still poor (Nödler et al. 2010, Schaffer et al. 2012a, Schaffer et al. 2012b, Hendriks 2013, Hillebrand et al. 2013, Niedbala et al. 2013, Miyajima 2012). As an example, despite the report of Mitchell et al. (1955) that the cationic dye methylene blue (MB) is poorly adsorbed onto iron-oxide-coated sand (thus, on iron oxides), Fe⁰ has been repeatedly reported to efficiently treat aqueous solutions containing MB (Ma et al. 2004, Frost et al. 2010). Actually this experimental observation cannot be refuted. However, it should be recognized that despite low affinity to iron oxides, enough adsorbents have been generated under experimental conditions to induce quantitative removal. This is exactly the principle of electro-coagulation (Bojic et al. 2009, Noubactep and Schöner 2010).

Another shortcoming of current procedures is the lack of characterization of used materials (Btatkeu et al. 2013 and ref. cited therein). For example, considering the work of Stieber et al. (2011), one should ask about the intrinsic reactivity of the commercial Fe^0 material (from G. Maier GmbH). If the intrinsic reactivity of this material is not known, what is the basis to compare the results to that obtained by other commercial materials (e.g. from Connelly-GPM, iPutec GmbH or Peerless Metals)? Moreover, other results are likely obtained only largely under different operational conditions (e.g. Fe^0 mass loading, mixing regime, volume of reaction vessels, pH value) (Miehr et al. 2004, Noubactep et al. 2009).

Another important feature from the Fe⁰/PPCPs literature is that conventional Fe⁰ (μ m, mm) has been first tested. For reducible PPCPs exhibiting low level of degradation by conventional Fe⁰, multi-metallic systems and nano Fe⁰ have been tested. However, as theoretically discussed (Noubactep 2009, Noubactep and Caré 2010a, Noubactep and Caré 2010b, Noubactep and Caré 2011, Noubactep et al. 2012b, Noubactep et al. 2012c) and experimentally demonstrated (Ghauch et al. 2011a, Gheju and Balcu 2011) the crucial factor is not primarily the higher reactivity of alternative materials but the (time-dependent) density of reducing agents within the system (Fe^{II}, H/H₂). This view has been recently independently confirmed by Liu et al. (2013) who introduced the concept of electron efficiency. Liu et al. (2013) have clearly criticized the current approach of material selection as a race to the most reactive material. The sole exception the author could identify is the work of Li et al. (2010),

purposefully alloying Fe^0 to decrease the kinetics of iron corrosion. Liu et al. (2013) introduced an economy-based factor of electron efficiency defined as the percentage of electrons utilized in a target reduction over an entire consumption at a given time interval. The concept of electron efficiency demonstrates that not the most reactive but rather the most appropriate material should be sought (Btatkeu et al. 2013).

The presentation above has clearly demonstrated that the research community on Fe^{0} remediation has being working on a false premise for more than two decades. To the best of the author's knowledge, only five active research groups have realized and addressed the original mistake: (i) the group of Dr. Gheju at the university of Bucarest (Romania), (ii) the group of Dr. Moraci at the university of "Reggio Calabria" (Italy), (iii) the group of Dr. Noubactep at the university of Göttingen (Germany), (iv) the group of Dr. Scott at the university of Bristol (United Kingdom), and (v) the author's group at the American University of Beirut (Lebanon). In essence, these five research groups are far from the still active 'circle of founding fathers' (Canada/USA) (Painter 2004, 2005). Over the years, several researchers have challenged to current premise (e.g. Lipczynska-Kochany et al. 1994, Odziemkowski et al. 1998, Lackovic et al. 2000, Lavine et al. 2001, Mantha et al. 2001, Furukawa et al. 2002, Jiao et al. 2009). However, their contributions were mostly limited at 'gently' or 'collegially' criticizing the validity of some aspects of the still prevaling reductive transformation paradigm. On the contrary, Noubactep (2006a, 2006b, 2007, 2008) clearly challenged the concept as a whole and suggested a paradigm shift (Noubactep 2011b, 2011c, 2013a, 2013b, 2013c). This has been accomplished with more than 70 peer-reviewed articles including (http://www.uni-goettingen.de/de/8681.html _ access 2013/07/04): (i) 23 short communications, 18 comments on research articles, (iii) 12 review articles and (iv) a thesis (habilitation degree). It is important to notice that the replies to Noubactep's comments were mostly severe, but not objective (Elsner et al. 2007, Tratnyek and Salter 2010, Nagpal et al. 2010). The next section will summarize the contribution of the author's group to the establishment of the state-of-the-art knowledge on processes yielding contaminant removal in Fe⁰/H₂O systems.

3.3 Removal of PPCPs in Fe⁰/H₂O systems: Beyond the state-of-the-art

As previously stated, the author contribution can be divided into two groups: (i) papers before P3 in which the experimental design and results interpretation followed the still prevailing reductive transformation paradigm ('dancing in the round'). P4 belongs to this group although

it was published later (2009), and (ii) papers after P3 which should be regarded as the major contribution to the validity of the alternative adsorption/co-precipitation paradigm.

3.3.1 Dancing in the round

The titles of the three papers from this group reveal that the presentation followed the reductive transformation concept (Ghauch 2008a, Ghauch 2008b, Ghauch and Tuqan 2008, Ghauch and Tuqan 2009): "*Rapid removal of flutriafol from water by zero valent iron powder*", "*Catalytic degradation of chlorothalonil in water using bimetallic iron-based systems*", and "*Reductive destruction and decontamination of aqueous solutions of chlorinated antimicrobial agent using bimetallic systems*". The critics made to the still prevailing paradigm are guilty for these three papers. However, the good documentation of reaction products was an excellent aspect in all three articles and are still valuable for the discussion of the efficiency of processes occurring in Fe⁰/H₂O systems.

The major corrections made for further works included: (i) the use of lower Fe^0 mass loadings, (ii) the use of various mixing devices and mixing intensities (including undisturbed systems), (iii) the use of MnO₂ to control the availability of corrosion products, (iv) the record of the pH value and the iron concentration during the experiments, and testing trimetallic systems.

It is worth noting that the author was initially sceptic about some aspects of Noubactep's model as can be read in the discussion of P3. For example the following paragraph is very illustrative:

"Dr. Noubactep has demonstrated that "contaminant coprecipitation through corrosion products has been shown to be primary removal mechanism in Fe^0-H_2O systems". This assumption could be true for a significant number of contaminants especially those without any halogens (X) on their structure. Because R-X bonds are sensitive to ET (electron transfer), this will lead to a breakdown of halogen bond and the halogens remain free into the solution unless they are adsorbed onto iron oxidation products."

This text passage illustrates the fact that the well-documented reductive transformation of contaminant could be the reason for the rejection of the co-precipitation concept. In fact adsorption and co-precipitation as fundamental removal mechanisms may be misinterpreted as negation of chemical reduction. Instead of simply refuting Noubactep's concept like other working research groups (Elsner et al. 2007, Tratnyek and Salter 2010, Nagpal et al. 2010), the author has poised to test its validity as it will be shown in the next section.

3.3.2 In tune with the mainstream science

To test the validity of the concept that contaminant are fundamentally removed from the aqueous phase by adsorption and co-precipitation (Noubactep 2007, 2008, 2010, 2011b), whether they are reduced or not, amoxicillin, ampicillin, β -lactam antibiotics, clofibric acid and diclofenac were used as probe PPCPs (Ghauch et al. 2009, Ghauch 2010, Ghauch et al. 2010a, Ghauch et al. 2010b, Ghauch et al. 2011a, Ghauch et al. 2011b). Used Fe⁰ materials included mm, µm and nm Fe⁰, bimetallic and trimetallic systems.

Results generally confirmed and completed the state-of-the-art knowledge on the effect of various experimental parameters. In particular: (i) the positive effect of controlled plating to form mono-, bi- and trimetallic systems, (ii) increased reduction under anoxic conditions coupled to a slower reaction kinetics, (iii) accelerated contaminant removal under oxic conditions. These well-documented aspects will not be addressed here, interested reader are referred to individual papers and works of Jiao et al. (2009), Gheju and Balcu I. (2011), Scott et al. (2011) or Btatkeu et al. (2013). The present overview will be limited to the major innovative contributions of the appended papers for the area of 'Fe⁰ remediation'.

3.3.2.1 The Fe⁰/MnO₂/H₂O system

A major innovation of the experimental design was the parallel investigation of $Fe^{0}/H_{2}O$ and $Fe^{0}/MnO_{2}/H_{2}O$ systems. These systems were reported in the literature for the investigation of radium, uranium and methylene blue removal in $Fe^{0}/H_{2}O$ systems (Burghardt and Kassahun 2005, Noubactep et al. 2005, Noubactep et al. 2006, Noubactep 2009). Published results were not univocal as the reported delay of contaminant removal was seemingly in contradiction with the claims that MnO_{2} sustains systems efficiency. Using more reactive bimetallic and trimetallic systems has resolved this issue (Ghauch et al. 2011a). In fact, MnO_{2} delays the availability of corrosion products but definitively sustains iron corrosion, which guarantees contaminant removal as reduction is just a side aspect. In particular, in column studies iron corrosion products improve adsorptive size-exclusion (Caré et al. 2013).

3.3.2.2 Oxic vs. anoxic conditions

Oxic vs. anoxic experiments for bi- and trimetallic systems have demonstrated the importance of material intrinsic reactivity and operational conditions for the extent of reduction, the nature of reaction products, and the detection of both educts and products (Ghauch et al. 2011a). In fact, for the same experimental duration, using the same mass loading has resulted in various extent of (i) contaminant reductive transformation, (ii) species co-precipitation, and

(iii) various intermediate reaction products. These results suggest that without an unified protocol for the investigation of processes in Fe^0/H_2O systems, produced data will keep a qualitative nature. For a proper design however, reliable quantitative data are needed.

3.3.2.3 Passivation of Fe⁰ materials

For the first time within the Fe⁰ remediation, experimental evidence is given demonstrating the importance of the kinetics of Fe⁰ corrosion for the passivation of its surface. As summarized by Nesic (2007), the protectiveness of an oxide scale on Fe⁰ depend primarily on the precipitation rate: "As the Fe⁰ surface corrodes under the scale, corrosion continuously undermines the scale. As voids are created, they are filled up by the ongoing precipitation, etc. When the rate of precipitation at the Fe⁰ surface exceeds the rate of corrosion (scale undermining) dense protective scales form. Vice versa, when the corrosion process undermines the newly formed scale faster than precipitation can fill in the voids, a porous and unprotective scale forms".

Working with trimetallic systems under oxic and anoxic conditions has corroborated this model. In fact the most reactive trimetallic system was the one who lost more rapidly his reactivity under oxic conditions suggesting that it was rapidly covered with an impervious layer of oxides. This result clearly shows that appropriate Fe^0 materials for site specific use should be identified/developed. A Fe^0 material which is very reactive will be rapidly passivated under more aggressive conditions. Here a less reactive material will be suitable. Therefore, a unified protocole for material testing is urgently needed.

3.4 Significance for future research

The observed reduction of oxidized contaminants and the oxidation of Fe^0 is not a simultaneously coupled-redox reaction. Clearly, contaminant reductive transformation is not the cathodic reaction coupled to anodic Fe^0 oxidative dissolution. In other words, although Fe^0 oxidative dissolution is an electrochemical reaction, contaminant reductive transformation is mostly a chemical reaction (electrons from Fe^{II} and/or H/H_2). However, as a rule the remediation industry is interested in removing contaminant and reduction of organics is mostly a mean to facilitate biodegradability. Whether the goal is reduction or elimination, the contaminant of concern has to reach the reaction site, the surface or the vicinity of Fe^0 .

The remediation industry has to deal with more than 100,000 substances at numerous sites with countless specificities (Hendriks 2013). Threatening more than 100,000 species requires an intelligent approach. A relevant strategy for the Fe^0 remediation industry could consist in assessing the affinity of relevant substances for iron corrosion products in laboratory

experiments. Later field studies could include multi-species settings. It is certain that this approach could be limited by financial constraints. However, primary target working with some relevant groups of species (e.g. anionic, cationic, neutral dyes) (Saha et al. 2011) would set the basis for a systematic research which can be first based on documented properties of individual species. In this approach lab and field experiments will be designed to fine tune design tools (thickness of the reactive bed, number of beds, mass of Fe⁰/bed).

4. Nanoscale and microscale iron particles for the improvement of the Fenton process under sonicated and silent systems

4.1. Introduction

Fenton's reagent is one of the oldest advanced oxidation technologies able to oxidize and destroy organic molecules under acidic conditions. The reaction products in most of the cases are benign (Diya'uddeen et al. 2012). The effectiveness of this technology toward water micro-contaminants has been demonstrated in a myriad of scientific publications and field applications during the last decades. In the conventional process, contaminated aqueous solutions are supplied with hydrogen peroxide (H₂O₂) and dissolved Fe^{II} species (from Fe salts). The whole process occurs in the aqueous phase (homogeneous reactions). The present work advocates for an alternative heterogeneous system (Segura et al. 2013). The Fenton process is coupled to low frequency ultra-sonication in an heterogeneous system in which dissolved Fe^{II} species are produced in-situ from the oxidative dissolution of micrometric or nanometric iron particles (Fe⁰ or nano-Fe⁰). For this purpose, the fundamentals of Fenton's reaction (advantages and limitations) will be presented first, then an overview on sonication at low frequency will be given, finally a summary of the results of carbamazepine and sulfamethoxazole degradation by Fenton/sonicated systems will be discussed.

4.2. Fundamentals of Fenton chemistry

Invented in 1890s by Henry John Horstman Fenton, the Fenton's reagent is developed in an aqueous solution containing H_2O_2 and Fe^{2+} species (Fenton, 1894). H_2O_2 is a weak acid (pKa = 11.65) not able to oxidize directly, at a moderate load, high concentrations of organic compounds with aromatic moiety. However, if mixed with transition metals like Fe^{2+} , the mixture becomes extremely high oxidative due to the formation of hydroxyl radicals OH^{\bullet} (HR, redox potential of about 2.80 V) upon H_2O_2 activation and this in a stoichiometric manner. In such systems, HRs are the primarily reactive species that harshly oxidize organic molecules classified as persistent organic pollutants (POPs) (Gan et al. 2009). HRs are known for their potential to add hydroxyl substituents to aromatic rings yielding therefore ring opening and, under optimum condition, full mineralization by converting POPs to carbon dioxide and water (Rogers and Bunce, 2001).

4.2.1. Mechanisms

Originally applied to homogeneous media, Fenton's reaction showed also its performance in heterogeneous systems where solid additives like iron oxides were able to catalyze the scission of H_2O_2 followed by the production of HRs. Figure 2 shows the schematic of organic compound oxidation by Fenton reaction. Many side reactions are involved in this process as widely reported in the literature and include: radical initiation, radical propagation and termination, reaction with intermediates, carbonate species and natural organic matter (NOM) (Eq. 8 to 27) (Kang and Hua 2005, De Laat and Lee 2005, Venny and Gan 2012).



Figure 2: Schematic of organic compound oxidation by Fenton reaction.

| $H_2O_2 + Fe^{2+} \rightarrow Fe^{3+} + OH + OH^-$ | (8 | 3) |
|--|----|----|
|--|----|----|

$$Fe^{3+} + H_2O_2 \rightarrow Fe^{2+} + HOO^{\bullet} + H^+$$
(9)

$$OH + RH \rightarrow R' + H_2O$$
(10)

$$R' + H_2O_2 \rightarrow ROH + OH$$
(11)

| $OH + Fe^{2+} \rightarrow Fe^{3+} + OH^{-}$ | (12) |
|--|------|
| $OH + OH \rightarrow H_2O_2$ | (13) |
| $\mathrm{Fe}^{2+} + \mathrm{HOO}^{\bullet} \longrightarrow \mathrm{Fe}^{3+} + \mathrm{HOO}^{\bullet}$ | (14) |
| $\mathrm{Fe}^{3+} + \mathrm{HOO}^{\bullet} \longrightarrow \mathrm{Fe}^{2+} + \mathrm{O}_2 + \mathrm{H}^+$ | (15) |
| $HOO' + H_2O_2 \rightarrow H_2O + OH + O_2$ | (16) |
| $HOO' + HOO' \rightarrow H_2O + 3/2 O_2$ | (17) |
| $O_2^{\bullet} + Fe^{3+} \rightarrow Fe^{2+} + O_2$ | (18) |
| $O_2^{\bullet} + Fe^{2+} + 2H^+ \rightarrow Fe^{3+} + H_2O_2$ | (19) |
| $HOO' + O_2' + H_2O' \rightarrow H_2O_2 + O_2 + OH'$ | (20) |
| $HOO' + OH \rightarrow H_2O + O_2$ | (21) |
| $CO_{2 (g)} + H_2O \leftrightarrow H_2CO_{3 (aq)}$ | (22) |
| $HCO_3^- + OH \rightarrow CO_3^- + OH^-$ | (23) |
| CO_3 + $H_2O_2 \rightarrow HOO' + HCO_3$ | (24) |
| 2 CO_3 \rightarrow Products | (25) |
| $CaCO_3(s) + H^+ \leftrightarrow Ca^{2+} + HCO_3^-$ | (26) |
| $NOM + OH \rightarrow Products$ | (27) |

The above equations are relative to reactions mainly occurring in homogeneous systems. However, in heterogeneous media using iron particles for example as H_2O_2 activators, additional reactions dealing with iron corrosion should also be considered. Reactive metallic iron (Fe⁰) for example produces, upon immersion, iron corrosion products (ICPs) including nascent oxides and Fe²⁺ that might play a crucial role in H_2O_2 activation (Eq. 28). The kinetics of iron corrosion certainly varies depending on the intrinsic reactivity of used Fe⁰ and the solution chemistry (O₂ level, pH value) (Fig. 3a). Concerning the variation as function of the acidity, the corrosion rate decreases with increasing pH till pH 4.0, becomes almost stable at 4.0 < pH < 10.0 and decreases slightly above pH 10.0 (Fig. 3a). Within a medium rich in ICPs, Fe²⁺ recycling might also occur in the presence of Fe³⁺ species (Eq. 29). However, since the solubility of Fe³⁺ is very limited in aqueous solutions of pH > 5 (Fig. 3b), this phenomenon is not really favored. Instead, Fe²⁺ might be regenerated after reaction of radical species with soluble Fe³⁺ (Eq. 30).

 $Fe^0 + 2H^+ \rightarrow Fe^{2+} + H_2$ acidic medium (28)

$$Fe^{0} + 2 Fe^{3+} \rightarrow 3 Fe^{2+}$$
 (29)

$$\mathbf{R}^{\bullet} + \mathbf{F}\mathbf{e}^{3+} \to \mathbf{R}^{+} + \mathbf{F}\mathbf{e}^{2+} \tag{30}$$


Figure 3. (a) Relative corrosion rate of iron as a function of pH (Wilson, 1923). It is arbitrarily assumed that at pH 4 iron corrodes with 12% of its rate at pH 0. (b) Solubility data of Fe^{II} in 0.1 M NaCl (25 C) and Fe^{III} in 0.01 M NaCl (25°C) as a function of pH (Noubactep and Schöner, 2010).

4.2.2. Types of Fenton's reaction

Three kinds of Fenton reactions have been described and tested for micro-pollutants: the Classical Fenton (CF), the Fenton-Like (FL) and the Amended Fenton (AM). To these three Fenton processes, one should add an improved Fenton-like oxidation process (IFOP) based on the use of low frequency ultra-sonication in heterogeneous solutions fed by H_2O_2 and micrometric or nanometric iron particles to form US/Fe⁰/H₂O₂ systems. The major challenge in improving Fenton process is the possibility to work at circumneutral pH instead of an acidic pH and to avoid the continuous feeding in H_2O_2 activators. CF is only based on dissolved Fe²⁺ however FL uses Fe³⁺, native or synthetized iron oxides or certain transition metals. Examples of iron oxides include: goethite (α FeOOH), hematite (α Fe₂O₃), magnetite (Fe₃O₄) and ferrihydrite (α Fe₁₀O₁₅•9H₂O). The AM, however, is based on the addition of some chelating agents preventing thereby iron precipitation even at neutral pH operating conditions. If very abundant in solution, H₂O₂ activators might alter the Reaction Stoichiometric Efficiency (RSE) of the reaction where huge amount of HRs is activated while few moles of contaminants are degraded. The catalytic conversion of H₂O₂ into HRs is so rapid that radical-radical quenching might occur.

In order to overcome such situation, an AM system called IFOP is proposed in which iron particles are the promoters of Fe^{2+} , Fe^{3+} and iron corrosion products (ICPs) into the solution that might be used for the production of HRs upon reaction with H₂O₂. The system became more efficient once coupled to low frequency US as it has been proposed for some permeable

reactive barriers for field application. This was helpful in cleaning iron particles surface making them more accessible by organic contaminants increasing thereby the longevity of the PRBs and keeping good efficiency in removing contaminants.

4.2. Fundamentals of ultra-sonication chemistry

Ultra sonication occurs at any wave frequency greater than 20 kHz. The ultrasonic wave includes compression and expansion cycles called rarefraction (Thompson and Doraiswamy 1999, Adewuyi 2001) during which dissolved molecules are entrapped by micro-bubbles that grow and expand under acoustic cycles. Upon adiabetic collapse, implosion of micro-bubbles occurs (Fig. 4). This will result in the release of high temperatures and high pressures yielding in some cases thermal dissociation of water molecules into H[•] and OH[•]. This phenomenon is known by the sonochemistry of water involving several chemical reactions (Eqs. 31-35).

$$\begin{array}{ll} H_2O +))) \rightarrow OH + H \qquad (31) \\ OH + H \rightarrow H_2O \qquad (32) \\ H + O_2 \rightarrow HO_2 \qquad (33) \\ 2OH \rightarrow 2H_2O_2 \qquad (34) \\ 2HO_2 \rightarrow H_2O_2 + O_2 \qquad (35) \end{array}$$

HRs produced during the cavitation process can diffuse to the interfacial regions and bulk solution encountering specifically hydrophilic and non-volatile compounds for further oxidation reaction yielding hydroxylation, ring opening, etc. until partial or complete mineralization of the organic target (David 2009).



Figure 4. Growth and implosion of cavitation bubble in aqueous solution with ultrasonic irradiation (Suslick 1989).

4.3. Coupling sonochemistry with Fenton using Fe⁰ (IFOP)

The developed IFOP is based on coupling of ultrasonication with Fenton reagent, however, using iron particles instead of dissolved iron species. The advantage of such system remains in the possibility of releasing smoothly ICPs responsible of the activation of H_2O_2 at low frequency sonication e.g. 20, 40 kHz. The latter is able to split H_2O_2 added into two HRs under the hot spot conditions in ultrasonic cavitation as the dissociation energy for the O—O bond in H_2O_2 is only 213 kJ/mol compared to 418 kJ/mol, the dissociation energy of O—H bond in H_2O (Eq. 36). In addition, US can improve iron corrosion by avoiding the formation of iron oxides able to stick at the surface of iron particles and consequently rendering passive the oxide scale at the surface of iron particles. At low frequency sonication, the amount of H_2O_2 produced is not at all significant to induce degradation of micro contaminants. Accordingly, the addition of H_2O_2 as external oxidant can greatly accelerate the degradation process especially if H_2O_2 is not in excess (Goel et al. 2004). Otherwise H_2O_2 can play a scavenger role after reaction with HRs producing 'OOH (weaker oxidant and much less reactive) (Eqs. 36-40) (Ku et al. 2005; Nam et al. 2003).

$$H_2O_2 +))) \rightarrow 2^{\bullet}OH \tag{36}$$

$$OH + H_2O_2 \rightarrow OOH + H_2O$$
(37)

$$OOH + OH \rightarrow O_2 + H_2O$$
(38)

$$2^{\circ}OH \rightarrow H_2O_2$$
 (39)

$$2^{\bullet}OOH \rightarrow H_2O_2 + O_2 \tag{40}$$

Sonication of aqueous solutions containing H_2O_2 at 20 kHz and slightly acidic pH is not capable of producing HRs. However, this becomes possible at basic pH (10-12) because H_2O_2 is a very weak acid (Svitelska et al. 2004). These conditions are not relevant to natural waters and effluents making the process not applicable. Instead, feeding the solution with iron particles under low frequency sonication can easily induce the diffusion of Fe²⁺, Fe³⁺ and ICPs into the bulk solution through the interface iron particles/solution. It can also favor the dissociation of H_2O_2 at the surface of iron particles upon the presence of some catalytic sites. Such process can be considered as an improved Fenton system because CF, FL and AM can take place at the same time into the solution in parallel to iron surface cleaning from oxides and ICPs under US effect. Basically, iron particles corrode in aqueous solution and are converted into Fe²⁺ (Eqs. 41, 42) (Merouani et al. 2010) and even Fe³⁺ after further oxidation (Eq. 43).

$$Fe^{0} + 2H_{2}O \rightarrow Fe^{2+} + H_{2} + 2OH^{-}$$
 (41)

$$2Fe^{0} + O_{2} + 2H_{2}O \rightarrow 2Fe^{2+} + 2H_{2} + 4OH^{-}$$
 (42)

$$2Fe^{2+} + \frac{1}{2}O_2 + 2H^+ \to 2Fe^{3+} + H_2O$$
(43)

 Fe^{2+} and Fe^{3+} react separately with H_2O_2 as previously reported to form 'OH and 'OOH, respectively (Eqs. 8, 9). However, 'OOH resulting from the dissociation of the complex Fe– O_2H^{2+} at a very slow rate can be greatly enhanced upon US effect (Eq. 44). The latter can also dissociate iron(III) complex to produce HRs as it is has been reported in our work on carbamazepine in the case of IFOP (Eqs. 45, 46). A cycle mechanism can then be established after the isolated Fe²⁺ further reacts with H_2O_2 to generate HRs (Eq. 8).

$$Fe-O_2H^{2+} +))) \rightarrow Fe^{2+} + OOH$$
(44)

$$Fe^{3^+} + H_2O \rightarrow [Fe(OH)^{2^+}] + H^+$$
 (45)

$$[Fe(OH)^{2^+}] +))) \rightarrow Fe^{2^+} + OH$$
(46)

Finally, one of the advantages of using iron particles in such oxidation processes is the increase in the nucleation sites at the proximity of iron particle surface favoring the process of cavitation inception increasing thereby the probability of formation of hydroxyl radicals.

The IFOP developed in our study was tested on solutions containing carbamazepine (CBZ) and sulfamethoxazole (SMX). CBZ was tested with only micrometric iron particles while SMX was degraded using nanometric as well as micrometric iron particles (Ghauch et al. 2011c).

4.4. Degradation of PAMs by IFOP

4.4.1. Case of CBZ

The study on carbamazepine investigated the suitability of the improved Fenton oxidation process in rapidly degrading one of the non-chlorinated PAMs that has been recently detected in water resources in France used to produce water for human consumption. Experiments were performed at room temperature ($21 \pm 2 \text{ °C}$) and different pHs (3, 5, 7 and 9). Powerful oxidative hydroxyl radicals for CBZ oxidation were produced by H₂O₂ (12.5-100 mM) and Fe⁰ (0.44-3.57 mM) additives under low frequency ultrasonication (40 kHz). Results showed no CBZ oxidation in the presence of an °OH scavenger (MeOH), confirming the Fenton's nature of the process. Acid-washed Fe⁰ feeding showed increased CBZ oxidative degradation up to 85%. Recycling experiments demonstrated also the ability of Fe⁰ to maintain its capability of releasing Fe²⁺ into the media to promote °OH generation for CBZ degradation (> 80% after 4 cycles). This was ascribed to the cracking effect on the surface of iron particles preventing thereby the formation of passive oxide. The process was also efficient in natural waters containing different species (e.g. Na⁺, Ca²⁺, Mg²⁺, Cl⁻, SO4²⁻, HCO3⁻, NO3⁻, etc).

The OH radicals were shown as the primary oxidative species after being generated by H_2O_2 additives in the presence of small amounts of Fe⁰ acting as pseudo catalyst under low US frequency (40 kHz). Complete degradation of CBZ (42 μ M, 20 mL) occurred at pH 3 after less than 30 min of reaction while almost 90% CBZ disappearance was reached at pH 5 after 1h of reaction with 100 μ L of H_2O_2 additives and 200 mg of Fe⁰ load.

Comparison between one and multiple steps H_2O_2 addition did not show any significant difference after 1h of reaction. Recycling experiments showed clearly the ability of the US/Fe⁰/H₂O₂ system to be effective even after 4 successive cycles. However, the efficiency of the system vanished if non acid-washed Fe⁰ is used.

Intermediate oxidation products (by-products) were identified by LC/MS. Most of them are unstable hydroxylated-CBZ that disappeared at longer treatment durations after being attacked by hydroxyl radicals, a sign toward full CBZ mineralization.

Finally, the results on CBZ demonstrated that IFOP $(US/Fe^0/H_2O_2)$ is a promising technique for oxidative removal of non-halogenated pharmaceuticals under mild conditions. It allowed successful removal of CBZ dissolved in natural water, by its conversion into biodegradable transformation products which could be degraded later within a biological step.

4.4.2. Case of SMX¹

Sulfamethoxazole (SMX) solution was subjected to low frequency ultrasonication (US) at 20°C and pH 5 in US/nanoFe⁰/H₂O₂ and US/microFe⁰/H₂O₂ systems. Removal rates of SMX were compared under different reagents loads. Results showed that (i) SMX degradation depends on water matrix and (ii) is positively affected by the presence of carbonate species favoring iron corrosion that enhances H_2O_2 activation. Experiments showed also that nanoFe⁰ were not efficient in activating H_2O_2 in silent systems while microFe⁰ showed better efficiency. This was ascribed to the increase in the nucleation sites at the proximity of microFe⁰ surface favoring the process of cavitation/inception increasing thereby the probability of formation of hydroxyl radicals. The developed system showed clearly that advanced oxidation processes work better in a Fenton-like process by heterogeneous catalysis in US/microFe⁰/H₂O₂ system instead of nanometric iron slurry solution e.g. US/nanoFe⁰/H₂O₂ system.

A comparative study undertaken in silent and sonicated systems using μFe^0 or nFe^0 is presented below.

¹ Results on SMX were presented at the 245th ACS national meeting at New Orleans on April 9, 2013.

4.4.2.1. µFe⁰/H₂O₂ systems (Silent vs US)

In the absence of $H_2O_2 \ \mu Fe^0$ does not seem to show an appreciable difference in the degradation efficiency of SMX between silent and US systems. However in the presence of H_2O_2 it has been found that sonication actually inhibited the degradation of SMX when compared to similar experiments operating under silent conditions. A further advantage that the silent system has over the US system is that it shows a slower rate of consumption of H_2O_2 especially at the lower iron concentrations. This discrepancy between the two systems may be ascribed to the increased iron(II) production under US conditions which acts as a quenching agent for hydroxyl radicals.

4.4.2.2. nFe⁰/H₂O₂ systems (Silent vs US)

In the absence of $H_2O_2 nFe^0$ seems to lose its efficiency and reaches a steady state within 30 min of reaction. Furthermore there is no observable difference between the US and silent systems. However in the presence of H_2O_2 an increase in the degradation efficiency of ~ 10% was observed for the silent system; and a substantial increase in the degradation efficiency especially for high iron loads under US conditions. This seems to suggest that sonication has the ability to make nFe^0 more accessible by avoiding its agglomeration into the solution increasing thereby its corrosion rate which favors the activation of H_2O_2 in a moderate way. Finally one can notice the change in color of the nFe^0 solution from gray to yellow in the presence of H_2O_2 indicating that the nFe^0 particles are more oxidized only in the presence of H_2O_2 .

4.4.2.3. Identification of the transformation product

The improved advanced oxidation process of SMX showed the presence of one transformation product analyzed by HPLC-MS (Fig. 5). The study of the mass spectrum of this product analyzed in positive ESI mode depicted ring opening of the isoxazole moiety followed by hydroxylation of the open ring. The resulting transformation product $([M+H]^+$ at 272.2 m/z) got disappeared at longer treatment time suggesting mineralization of SMX into more harmless products.

4.5. Conclusion: nFe^0 vs μFe^0 in Fenton-like oxidation

Presented results clearly demonstrated the importance of systematic investigations for the establishment of the science of 'Fe⁰ remediation'. Considering the particle size alone, nFe⁰ systems have a clear advantage over μ Fe⁰ systems in terms of its consumption of H₂O₂. This result is in tune with increased reactivity of nano-particles (Comba et al. 2011, Crane and

Scott 2012, Noubactep et al. 2012c). However, nFe^0 systems do not exhibit excellent reusability results as is evidenced by the pronounced change in the color of the solution in the presence of H₂O₂. This result suggests rapid and full oxidation of iron nano-particles within the experimental duration. It is clearly demonstrated that μFe^0 is advantageous in the sense that it is not completely exhausted for the experimental duration. μFe^0 can be reused for numerous cycles of reaction, as well as the fact that it does not need sonication to be activated which is a good advantage in terms of energy consumption. This observation is the essence of using nFe^0 systems for rapid degradation but their long-term reactivity under environmental conditions has not been properly addressed (Noubactep and Caré 2010b, Crane and Scott 2012, Noubactep et al. 2012b). These fundamental aspects of iron corrosion should be routinely considered in all systems involving Fe^0 . Reproducible and comparable results will only be obtained when Fe^0 reactivity and all other operational parameters are properly considered.



Figure 5: (a) HPLC/MS chromatogram of SMX during Fenton process. (b) MS spectra of SMX and its transformation product.

5. Persulfate for the degradation of organic contaminants: Activation through thermal and chemical ways using zero-valent iron particles

5.1. Introduction

Persulfate is considered the newest oxidant being used for the decontamination of soil and groundwater (Tsitonak et al. 2010). A search in Scopus database (23/06/2013) using the keywords: 'oxidation', 'persulfate', and 'pollutants' revealed a total of 141 articles published during the last 3 decades. From the 141 articles, 90 (64 %) have been published since 2009. In other words, only 51 articles have been published in almost 25 years. These statistics clearly show increasing interest of the scientific community in persulfate based degradation processes. Further systematic investigations are needed in order to (i) elucidate the degradation mechanisms of relevant organic compounds and (ii) optimize system design relevant to field operations.

Initially used for organic synthesis (Austin 1911), polymerization (Dyer et al. 1956) and as digester in the total organic carbon (TOC) analyzers (Walker and Clifford 2002) for the determination of TOC in water samples and effluents (Sharp 1973), persulfate (PS) becomes in recent years one of the most used oxidants for environmental applications. A survey of the three named original applications reveals that these are technical operations performed under controlled conditions (by skilled personnel). In particular, PS needs an initial (or a primary) activation to be efficient. Accordingly, to be successfully used in environmental applications, intelligent systems for PS activation must be developed.

Activated PS generates sulfate radicals (SRs) of high redox potential ($E^0(SO_4^2/SO_4)$) = 2.40 V) (Huie et al. 1991). This E^0 value is comparable to that of hydroxyl radicals (HR) ($E^0(HO^-/HO^-)$) = 2.7 V) (Buxton et al. 1988). HRs have been widely used in environmental applications (see Chapter 4). However, those radicals are not very efficient in terms of selectivity and reaction stoichiometric efficiency (RSE). Therefore, catalysts capable of increasing lifetime of HRs precursors (e.g. H_2O_2) have been developed/synthesized in order to optimize the oxidation process in terms of longevity (Pham et al. 2009, Huang et al. 2013). In contrast, SRs are more selective and can be generated in a more controlled fashion from many PS salts precursors (e.g. Na^+ , K^+ , NH_4^+) for field applications over longer period resulting in sustainable decontamination solutions (Anipsitakis and Dionysiou 2004, Waldemer et al. 2007, Liang et al. 2008, Antoniou et al. 2010, Vicente et al. 2011, Rodriguez et al. 2012, Fang et al. 2013b).

This chapter briefly gives an overview on the state-of-the-art of activation methods for PS with emphasis on (i) thermal activation upon heating and (ii) chemical activation using micrometric iron particles (MIPs) and iron corrosion products (ICPs). Own results gained with several probe species are summarized to substantiate that the efficiency of the PS/Fe^{0} system is primary a function of the reactivity (e.g. the intrinsic properties) of the organic molecules of concern. Optimum conditions for degradation reactions are determined based on a maximum of RSE. As for the elucidation of the degradation mechanisms of each probe, different schemes are available based on the identified intermediates. Sulfonated intermediates were identified for the first time during PS thermal oxidation of methylene blue as a dye probe.

5.2. Fundamentals of Persulfate Chemistry

5.2.1. Non activated persulfate

Highly soluble in water, persulfate salts $(Na_2S_2O_8, K_2S_2O_8 \text{ and } (NH_4)_2S_2O_8)$ dissolve yielding strong however relatively stable PS anions according to the following equations:

$$K_2S_2O_8 \rightarrow 2 K^+ + S_2O_8^{2-}$$
 (47)
 $S_2O_8^{2-} + 2e \rightarrow 2 SO_4^{2-}$ (48)

A review by House (1962) compiled the major PS oxidation reactions with inorganic species as well as organic molecules. It also described the mechanism of PS oxidation reactions as well as some kinetics study relative to the rate of oxidation of some compounds by PS. His review stills providing basic information on PS oxidation reactions with almost 607 citations in web of science (July 2013). It serves as a support for the application of PS oxidation of PS oxidation to an important number of organic molecules of environmental concern.

5.2.2. Activated persulfate

Several ways have been considered for the activation of PS into SRs. All of them are based on the breakdown of the O-O bond. The corresponding uncatalyzed reaction requires activation energy of about 33.5 kcal per mole (Kolthoff and Miller, 1951, Kolthoff et al. 1951). Accordingly, a raise in the temperature of the solution allows such energy to be reached (Waldemer et al. 2007) or the use of a metal ion (Anipsitakis and Dionysiou 2004) can easily catalyze the breakdown of the O-O bond. Also, photolysis (Lin et al. 2011) and microwaves (Yang et al. 2009) showed potential activation of PS into SRs however under specific conditions. The produced SRs are species known by their very high reactivity and short lifespan able to oxidize with different extent most of the organic compounds much like HRs however under different mechanism. SRs remove electrons from an organic molecule to produce an organic radical cation, such as Eq. (49) (Forsey, 2004), while HRs abstract hydrogen from the C-H bond (Neta et al., 1977) or add to C=C bonds.

$$\mathrm{SO_4}^{\bullet-} + \mathrm{M} \to \mathrm{SO_4}^{2-} + \mathrm{M}^{\bullet}$$
 (49)

SRs are electrophilic reagents favoring high oxidation rates of aromatic organic molecules containing electron donating groups such as -(NH₂), (-OH), or (-OR). However, if electronwithdrawing groups (-NO₂, C=O) are present on an aromatic cycle, the oxidation rate of the reaction decreases (Neta et al. 1977). On the other hand, substituents as Cl atoms are also expected to slow oxidation rates. For example, reaction rates for halogenated benzoic acids were 10 times lower than those for benzoic acids (Neta et al. 1997) and the presence of halogens in organic compounds (Sedlak and Andren 1991, Haag and Yao 1992, Tang and Huang 1995) has been shown to slow down reactions with HRs, and it is possible that the same applies for SRs.

Furthermore, solution additives (matrix effect) might also have negative effect on the oxidation rates by SRs. In fact, some of the inorganic species omnipresent in water (e.g. $CO_3^{2^-}$, HCO_3^- , Ca^{2^+} , Mg^{2^+} , Cl^- , NO_3^- , $PO_4^{3^-}$, $H_2PO_4^-$, etc) react with SRs decreasing therefore the efficiency of the degradation reaction of organic contaminants. For example, the very well-known case is the one of carbonate species. Depending on the pH of the solution, these species might largely affect the degradation rate because of the reaction of SRs with carbonate species as supported by Eq. 50.

$$\mathrm{SO}_4^{\bullet-} + \mathrm{HCO}_3^{-} \to \mathrm{SO}_4^{-2-} + \mathrm{CO}_3^{\bullet-} \tag{50}$$

Finally, a decrease in the degradation rate of contaminants might take place in solutions of high ionic strength *I* that can be explained by a synergetic effect of all ions together into the solution yielding quenching phenomena of SRs. Accordingly, a case-by-case study could not reveal or reflect correctly and accurately the effect of separated ions into the solutions on the degradation rate of organic micro-contaminants. Instead, a comprehensive study should take into account all species in a mixture mimicking natural conditions.

Some of the useful methods for persulfate activation are summarized below.

5.2.2.1. Thermal activation

Upon heating (40-99°C), PS can be easily converted into SRs having one unpaired electron. Sulfate free radical is formed by thermal decomposition of the persulfate anion in an

aqueous solution as shown in Eq. 51. The reaction kinetics follows first order and is better catalyzed in solutions of acidic pH (House 1962).

$$S_2 O_8^{2-} \rightarrow 2 SO_4^{\bullet-}$$
(51)

Thermal activation is the simplest way to activate persulfate into SRs intermediates avoiding thereby any additives to the solution. It can be considered a green chemistry process because no chemicals are needed to achieve activation. However, thermal activation is an energy demanding process and its application to relatively high effluent volumes cannot be considered as cost effective by industrials. Accordingly, if applied, this should only be done on small effluents' volume like for example hot spot hospital effluents, pharmaceutical industries, PAH charged effluents (Zhao et al. 2013, Tan et al. 2012, Deng and Ezyske 2011, Huling et al. 2011, Antoniou et al. 2010, Gosetti et al. 2004). Another feature of using thermal activation is the possibility of its application to already heated effluent which is the case of some dyeing industries producing effluents at temperatures up to 50°C where no additional heat is needed. Recall that textile dyeing industries need huge quantity of water for textile dyeing, which they normally pump out repeatedly from the ground or natural water sources resulting in depletion of ground water level.

In this sense, using solar energy to heat normal effluents from room temperature to 40-50°C can be a good alternative in order to minimize the decontamination process cost. Such possibility is applicable in countries where abundance of sun is usually present e.g. (MENA, Gulf).

Appended articles P13 and P14 are some of the rare papers recently published (Nfodzo and Choi, 2011) and dealing with the degradation of pharmaceutical compounds using PS thermal oxidation. However, appended article P15 shows our contribution in the elimination of MB for it, sulfonated intermediates were identified to clearly demonstrate the direct implication of SRs in the oxidation process.

5.2.2.2. Chemical activation

In the presence of electron donor metals (e.g. Fe, Co, Ag, Mn, Cu, etc.) (Liu et al. 2012), PS can be chemically activated to produce sulfate free radicals at even room temperature (Eq. 52). The reaction is so rapid that excess of activators might allow SRs quenching by favoring radical-radical reactions (Eq. 53) instead of radical-organic molecule reactions.

$$S_2O_8^{2-} + Me^{n+} \rightarrow SO_4^{\bullet-} + SO_4^{2-} + Me^{(n+1)+}$$
 (52)

$$\mathrm{SO}_4^{\bullet-} + \mathrm{SO}_4^{\bullet-} \to \mathrm{S}_2 \mathrm{O}_8^{2-} \tag{53}$$

Accordingly, the reaction should be well controlled in terms of additives so as to optimize the oxidation process. Furthermore, the concentration of metallic species should be maintained below the MCL defined by regulator's bodies in order to comply with recommended standards values.

Chemical activation has also been tested using metallic oxides (synthetic and natural) at the micrometric and nanometric scale (Zhu et al. 2013), citrate chelated iron, alkaline solution, hydrogen peroxide (H_2O_2) (Zhao et al. 2013), quinones and humic acids (Fang et al. 2013), minerals e.g. siderite (Huang et al. 2013), ferrous hydroxide colloids (Yan et al. 2013), granular activated carbon (Huling et al. 2011), etc. However, there is no published data on the longevity of the oxidation process once iron particles are used to activate PS and the role that can play in-situ generated iron corrosion products (ICPs) in sustaining the oxidation process over longer periods.

In this sense, the last appended article (P16) shows our contribution in understanding PS chemical oxidation of sulfamethoxazole (SMX) that has been improved in heterogeneous medium. This improvement is due to the adsorption of SMX on the surface of nascent iron oxide whose chemical properties are much elaborated than commercial iron oxides.

5.2.2.3. UV light and sun light activation

Upon irradiation with UV light, PS can be converted into SRs without any specific additive making easier kinetics studies in such medium. Several papers appeared recently on UV activated PS. Applications to a wide range of pollutants was successful as for example the mineralization of 2,2,3,3-tetrafluoro-1-propanol (Shih et al. 2013), the destruction of cyanobacterial toxin cylindrospermopsin (He et al. 2013), the degradation of rhodamine B (Chen et al. 2012), the oxidation of sulfamethazine (Gao et al. 2012) and bisphenol A (Yoon et al. 2012), etc. According to these authors, UV PS activation is a promising advanced oxidation technology for the degradation of organic contaminants. Its application can be extended to solar activation reducing thereby the operational cost.

5.2.2.4. Ultrasonic (US) PS activation

It has been demonstrated that a PS solution can produce under low frequency US SRs to carry out, in aqueous solutions, oxidation reactions yielding degradation and mineralization of organic contaminants. Under acoustic cavitation, PS is able to split into 2 SRs as reported in Eq. 54.

$$S_2O_8^{2-} +))) \rightarrow SO_4^{\bullet-} + SO_4^{\bullet-}$$
(54)

Those radicals can undergo additional reactions (Eqs. 55, 56) to produce hydroxyl radicals as secondary oxidation source of organic compounds.

$$SO_4^{\bullet^-} + H_2O \rightarrow H^+ + SO_4^{2^-} + HO^{\bullet}$$
 (55)
 $S_2O_8^{2^-} + HO^{\bullet} \rightarrow HSO_4^{-} + SO_4^{\bullet^-} + \frac{1}{2}O_2$ (56)

However, if high frequency US is used, additional species can add to the oxidative medium enhancing thereby its efficiency in terms of chemical oxidation (Eqs. 57, 58).

$$H_2O +))) \rightarrow HO^{\bullet} + H^{\bullet}$$
 (57)

$$S_2O_8^{2-} + H^{\bullet} \to HSO_4^{-} + SO_4^{\bullet-}$$
(58)

This can also be accompanied by pyrolysis of PS upon raise of the temperature of the solution under sonication (Eq. 59).

$$S_2O_8^{2-} + (Pyrolysis) \rightarrow 2 SO_4^{\bullet-}$$
 (59)

5.2.2.5. Other PS activation methods

5.2.2.5.1. Microwave (MW)

In this case, microwaves are only used to heat the solution and therefore the result is similar to thermolysis where PS is converted to SRs by the mean of temperature. Some applications using MW-AOP are known as for example the discoloration of acid Orange 7 (Yang et al. 2009), the removal of 1,1,1-trichloroethane (Li et al. 2009) and oxidation of As(III) into As(V) in an aqueous environment (Neppolian et al. 2010).

5.2.2.5.2. Electron beam (EB)

Electron beam technology has also been tested on PS solutions in order to investigate PS activation into SRs. For example investigations into this direction included the irradiation of PS solution in the presence of different ions (Criquet et al. 2011), the oxidation of benzotriazole and humic acid (Roshani et al. 2011), and the influence of persulfate ions on the removal of phenol in aqueous solution (Boukari et al. 2011).

In all the above mentioned activation techniques, recently appeared the problem of secondary products formed upon oxidation as for example bromate in solutions containing bromide ions. This issue is seriously considered by the scientists and several studies with regard to this topic are under progress (Fang and Shang, 2012).

5.3. Application to PPCPs

Table 1 summarizes the most important physical properties of the different PPCPs investigated by PS oxidation through thermal (TA-PS) and chemical activation (CA-PS)

techniques. As it can be seen, these molecules have different properties giving them different affinities vis-à-vis of the advanced oxidation treatment in terms of degradation mechanism and resistance to mineralization. Only after proper consideration of these intrinsic properties of individual PPCP could the impact of additional parameters be traceably investigated. A paramount importance should be reserved to matrix composition of the contaminated water (pH, ionic strength). In fact, the presence of dissolved inorganic species might significantly affect the oxidation process. A general overview of these impacts have been investigated and presented in the appended articles P13-P16 (Ghauch and Tuqan 2012, Ghauch et al. 2012a, Ghauch et al. 2013).

| | Bisoprolol (BIS) | Ibuprofen (IBU) | Methylene Blue (MB) | Sulfamethoxazole (SMX) |
|-----------------------------|---|-------------------------------------|--|------------------------------|
| Medicinal property | β-blocker Hypertension | Non-steroidal Anti-inflammatory | Dye, monoamine oxidase inhibitor | Bacteriostatic Antibiotic |
| CAS Number | 66722-44-9 | 15687-27-1 | 7220-79-3 | 723-46-6 |
| Chemical Formula | C ₁₈ H ₃₁ NO ₄ | $C_{13}H_{18}O_2$ | $C_{16}H_{18}N_3SCl$ | $C_{10}H_{11}N_3O_3S$ |
| Molecular weight (g/mol) | 325.44 | 206.29 | 319.85 | 253.279 |
| Chemical structure | | CH ₃ H ₃ C | H_3C_N CH_3 $CI^ CH_3$ $CI^ CH_3$ | |
| Solubility (mg/L) | 2240 | 21 | 43,600 | 610 |
| рКа | 9.5 | 4.91 | 3.8 | 6.16 |
| LogP (lipophilicity) | 1.87 | 3.97 | 0.9 | 0.89 |
| Density g/mL | 1.033 | 1.029 | 1.0 | 1.462 |
| mp °C | 100 | 75-78 | 100-110 °C | 166-169 |

Table 1: Selected physical properties of pharmaceutical and personal care products (PPCPs) tested in this work.

5.3.1. Case of Bisoprolol (TA-PS)

Experiments undertaken on BIS solutions showed successful degradation over 1 h in aqueous solutions under thermal persulfate activation (40-70°C) through oxidation reactions

involving different radicals e.g. $SO_4^{\bullet-}$ and HO^{\bullet} . The calculated reaction stoichiometric (RSE) efficiency was very satisfying at 50° C (RSE = 1.0) and 60° C (RSE > 0.8). The reaction kinetics was attributed pseudo-first order rate with respect to BIS. Results showed that high temperatures and neutral pH values e.g. 7.0 were more favorable than acidic solutions and low temperatures for complete BIS degradation. Alcohol additives demonstrated the responsibility of hydroxyl radicals HO^{\bullet} as well as sulfate radicals $SO_4^{\bullet-}$ in oxidizing BIS. Among all inorganic additives, only HCO₃ slightly influenced the observed degradation rates of BIS. Tests done on spiked BIS drinking water showed almost complete BIS degradation over 1 h of reaction regardless of a slight decrease in the observed degradation rate. The study of the total ion chromatograms obtained showed explicitly the formation of intermediate BIS oxidation products through hydroxylation and/or peroxydation. Oxidation reactions extended over 2 h confirmed the total disappearance of all byproducts. However, in order to confirm full mineralization, additional work should be undertaken using appropriate analytical techniques. The efficient oxidation of BIS can be considered as good indicator in order to extend this AOP to additional pharmaceutical persistent molecules dissolved in water or adsorbed on soils.

5.3.2. Case of Ibuprofen (TA-PS)

This work demonstrated that thermally activated persulfate can also effectively oxidize recalcitrant pharmaceuticals like IBU over a reasonable period of time (e.g. min to hours) at pH 7.0 with high RSE of about 0.65 compared to that of pHs 4.0 and 9.0. The oxidation reaction showed excellent Arrhenius behavior with activation energy of about 168 (±9.5) kJ mol⁻¹ higher than the one previously obtained with BIS e.g. 119 (±10.8) kJ mol⁻¹ which confirm the more recalcitrant character of IBU compared to that of BIS. Reactions in the presence of radical quenchers (EtOH and TBA) showed the implication of both radicals HO^{\bullet} and $SO_4^{\bullet\bullet}$ in oxidizing IBU. Results showed also that IBU degradation can be explained by pseudo-first order model as for BIS however a deep investigation into the reaction order yielded a more accurate kinetics rate equation that can be described by the following equation: $-\frac{d[IBU]}{dt} = (2.71x10^{-4}mM^{1-(a+b)} min^{-1})x[IBU]^{0.25}[SPS]^{1}$. The oxidation reaction

exhibited a fractional order 0.25 with respect to IBU demonstrating that a change in $[IBU]_0$ at a fixed $[SPS]_0$ would result in different IBU degradation rates. Furthermore, the elaborated kinetics model showed that the reaction is first order with respect to SPS therefore an increase

in $[SPS]_0$ for a fixed $[IBU]_0$ will result in a faster IBU degradation rate. Future attempts are planned in order to ckeck the validity of this kinetics model to similar pharmaceutical molecules serving as acting ingredient in drug formulation. This could be considered as additional evidence of the feasibility of the thermal activation PS process that can be used in order to establish a kinetics model widely applied for waste water treatment charged in pharmaceuticals.

5.3.3. Case of Methylene Blue (TA-PS)

Experiments carried out on MB solutions showed clearly full discoloration and important degradation of the dye by thermally activated SPS at different temperatures (30-70°C). Results showed that discoloration kinetics depend on the initial concentrations of MB and SPS, the pH and the activation temperature as well. The reaction kinetics fits well with the pseudo-first order model ($R^2 > 0.99$), however, under specific conditions where the molar ratio $[SPS]_0/[MB]_0 > 32$. Under non-moderate conditions (50-70°C), full MB discoloration occurred quickly. This was totally achieved in less than 20 min for $[SPS]_0/[MB]_0 \ge 640$. However, under moderate conditions (30-40°C) running over two weeks, complete discoloration is reached in less than one day for $[SPS]_0/[MB]_0 \sim 160-1,600$ and in 4-10 days for $[SPS]_0/[MB]_0 \sim 16-64$. On the other hand, MB discoloration showed an Arrhenius behavior with greater activation energy for solutions of $[SPS]_0/[MB]_0 < 32$ than for solutions of $[SPS]_0/[MB]_0 > 32$ (e.g. $E_A = 145.3 (\pm 27.5) \text{ kJ mol}^{-1}$ and 107.4 $(\pm 24.4) \text{ kJ mol}^{-1}$, respectively). This correlates with the principle of the pseudo-first order kinetics where the concentration of the oxidant should remain in large excess compared to that of the treated molecule. MB degradation by thermally activated SPS showed also an advantage of being successful for the dissolution and the degradation of MB-derivative precipitate formed after the addition of SPS to MB solutions at moderate temperatures. Both sulfate and hydroxyl radicals are responsible for the oxidation of MB into colorless derivatives. The LC-APPI-MS analysis showed the ephemeral character of those MB sulfonate derivatives. In fact, after 20 min of reaction, complete disappearance of all intermediates was noticed. This can strongly be considered as a possible route to full mineralization making from thermally activated SPS an advanced oxidation process that can be successfully used for the treatment of industrial effluents contaminated with dyes. Heating effluents to above 40°C is a costly process that is hardly affordable by industrials. However, this could be minimized if solar stills are used to increase the temperature of those effluents to more than 40°C accelerating therefore the discoloration process. Research in this direction is currently in progress.

5.3.4. Case of Sulfamethoxazole (CA-PS)

This work demonstrated that PS chemical activation is a reliable method for the treatment of water contaminated with recalcitrant SMX (39.5 μ M) in less than 2 h. Fe⁰/H₂O systems were efficient in activating PS progressively minimizing therefore under optimized conditions SRs quenching. A small load of commercial micrometric iron particles (MIPs) (e.g. 2.23 mM) was sufficient to completely degrade SMX over a 60 min reaction time. The application of PS in different water-based matrices revealed that initial pH is important to consider, especially in the context of initiating iron corrosion responsible for the release of Fe²⁺ species for PS activation. Results using deionized (DI), tap and underground (UG) water revealed similarities between the last two water types where SMX degradation efficiency was directly affected by bicarbonate content. In contrast, after its elimination through acidification, all solutions showed similar behavior toward SMX degradation over 3 cycles of 1 h until complete MIPs depletion. SMX degradation in Fe⁰/H₂O systems showed the presence of a stable transformation product. However, in Fe⁰/H₂O PS-spiked systems, complete disappearance of SMX and its transformation product occurred; the mineralization extent reached 37% under smooth conditions making PS oxidation technology an excellent method for the treatment of organic micro-contaminants. Additional investigations into the long-term use of PS activators like bimetallics, trimetallics and natural iron oxide-based systems (e.g. ferrihydrite, lemonite, etc.) might be useful in order to improve the RSE that reached a maximum of 5.2% in the current work (Ayoub and Ghauch, 2014). Compared to thermal activation, PS chemical activation could present an economical advantage in terms of energy consumption and long term application if natural minerals could demonstrate their effectiveness for sustainable solution toward PS activation. Research in this direction is in progress in our laboratory for a comprehensive optimization of PS/Activators in aqueous systems using natural minerals.

5.4. Conclusions

The use of PS as oxidant has proved its effectiveness in degrading PPCPs under very moderate conditions that can be encountered in real case effluents. For example, the pH, ionic strength, ions content, concentration of PS, temperature and activator loads are parameters that can affect the efficiency of the oxidation process.

Iron particles demonstrated high potential of PS activation especially for long term applications because of their smooth oxidation in the medium. Iron corrosion products (ICPs)

generated include dissolved Fe(II) and nascent iron oxides capable of sustaining the oxidation process due to the intrinsic properties of those ICPs known by their high reactivity.

Optimization of operational parameters has been established. For example the RSE reached in TA can lead to 100% efficiency increasing thereby the longevity of the process. Also, a kinetics model has been developed with possible implementation to specific effluents of small volumes e.g. hospital effluents.

The results showed also for some species the implication of both SRs and HRs in the oxidation process. The identification of sulfonated transformation products can be considered as an evidence of the direct implication of SRs in the oxidation process and their reactivity toward aromatic cycles. However, hydroxylated intermediates demonstrated in the majority of the cases the implication of HRs in the oxidation process.

The mineralization extent was acceptable (e.g. 37%) considering the small concentration of PS used (1 mM). This could be much higher if greater PS amount has been used e.g. 20-100 mM. However, this should also be well controlled in order to avoid radical quenching and therefore a decrease in the RSE of the reaction.

5.5. Future developments

In view of the obtained results, several recommendations can be addressed in order to better improve the use of PS for the degradation of recalcitrant organic molecules especially PPCPs.

For example, more emphasis should be given to natural activators usually abundant without any prior treatment minimizing therefore the operational cost. Some of these activators are iron-based systems in the form of oxides or silicates to be collected in zones that was under volcanic activities.

Another source of PS activators can be found in several industrial wastes as for example the iron collected upon cleaning of the surface of rotor disk before the installation of new pad breakers. Preliminary results (not yet published) indicate clearly the potential of this iron waste in activating smoothly PS in aqueous solutions of circumneutral pH.

Coupling CA with TA might increase the mineralization yield of organic contaminants because of the high density of SRs that will be produced in the medium. Despite the fact of being expensive if classical energy is used to heat solutions for PS oxidation, renewable energy can be used for this goal. Existing or new engineered solar stills can help in this direction because temperatures of 50° C can be easily reached.

Finally, it also might be interesting to do column experiments in which a reactive zone of a mixture of sand and iron (to define the % of each) is used for an influent gradually spiked by PS. In such a case, permanent oxidation can be reached accompanied by water filtration on sand/iron oxides bed.

6. Concluding remarks

Remediation with elemental iron (Fe^0 or ZVI) is overviewed in this thesis. Issues and limitations of experimental designs aiming at optimising the Fe^0 technology were identified and avenues for further improvements suggested. At the term the huge potential of this excellent water treatment material could be fully exploited in appropriate designs.

Remediation with Fe⁰ is a rapidly growing sector of environmental science. A search at Scopus (23/06/2013) with 'zerovalent iron' and 'remediation' revealed that 2,029 scientific articles may have been published on this sector since 1994. In particular, a growing body of evidence has proven Fe^0 as highly efficient and affordable remediation material. In recent years, the focus was mainly directed on nano-scale Fe⁰ and various Fe⁰-based systems (including bimetallic and persulfate heated systems). However, the general competiveness of Fe⁰ against conventional treatment systems (e.g. adsorption on granular activated carbon, chemical oxidation) is still widely disputed. It is even not yet established, whether Fe⁰ plays any significant role in the process of contaminant chemical reduction. To sum up, the available literature on remediation with Fe⁰ is a confusing array of independent data (Henderson and Demond 2007, Comba et al. 2011, Ghauch et al. 2011, Gheju 2011, Crane and Scott 2012, Noubactep 2012, Noubactep et al. 2012b, Btatkeu et al. 2013, Tepong-Tsindé et al. 2015a, Tepong-Tsindé et al. 2015b). Consequently, in order for different Fe⁰-based systems to be effectively contrasted with each other and to other conventional remediation technologies, a universal empirical testing framework is required. In particular, the term 'reactivity' should be carefully used and its relationship to 'efficiency' well-defined (Miyajima 2012, Btatkeu et al. 2013, Miyajima and Noubactep 2015).

The reactivity of Fe⁰ material is commonly characterized in short term (lab or field) experiments. The initial reactivity is typically high as aqueous iron corrosion is a self-inhibitive process. Therefore, the efficiency of a Fe⁰ system should be designed based on the 'residual' material reactivity: that is the Fe⁰ reactivity beyond the formation of an oxide scale. Fig. 6 illustrates the evolution of the treatment efficiency of two different Fe⁰ systems as compared to a reference (non-reactive system). It is seen that the reference system is not reactive at all. Here, the breakthrough occurs after 10 % of the treatment duration (arbitrary relative value). Such a system can be sand or pumice filter (media filtration). The two other systems could be differentiated by the nature of used Fe⁰ (nm, μ m or mm) or their amounts. Fig. 6 reveals that for up to 22 % of the treatment time both systems are very efficient at treating the water of concern. An initial breakthrough is observed in system ZVIa after some

22 days and pseudo-equilibrium after some 30 days. This pseudo-equilibrium corresponds to a relative removal efficiency of about 50 %. In other words, system ZVIa is suitable for water for which 50 % removal of the initial amount of contaminant is satisfactorily. If the extent of contaminant removal should reach 70 to 80 %, then system ZVIb is the suitable material to be used. It should be pointed out that ZVIa can still be used to achieve 70 to 80 % contaminant removal, but in a different design (e.g. two columns in series, a low flow rate).



Figure 6: Schematic representation of the time-dependant evolution of the efficiency of three filtration systems: a non-reactive system (reference) and two Fe⁰-based systems.

The discussion in Fig. 6 is based on a conventional granular Fe^0 material for reactive filtration. However, the resulting rule of thumb can be extended to all other Fe^0 -based systems. Thereby, the main question is 'how long should the used material be efficient?' For nano-Fe⁰ for example, the total reaction time could be some few days/weeks, while the duration for micro-Fe⁰ could reach some years/decades. Similarly, for wastewater treatment a micro-Fe⁰ can be recommended for use in some treatment cycles before recycling. Here, recycling could just mean 'allow the surface layers to be corroded by air O₂' ('initiation/regeneration by aeration'). It is essential to recall that the kinetics of Fe⁰ corrosion is a paramount design parameter as water is treated by corrosion products (not by the parent Fe⁰). The validity of this assertion is more evident for other Fe⁰-based systems (e.g. Fe⁰/H₂O₂, Fe⁰/US/H₂O₂, Fe⁰/Persulfate) than for reactive filtration (Fe⁰/H₂O system).

The universal acceptance of Fe^0 remediation systems will not occur until a fundamental understanding of involved processes, interactions and impact has been traceably

demonstrated. Therefore, future research must seek to establish a sound body of evidence upon which predictions of Fe^0 service life may be accurately made. For reactive filtration systems, such evidence has already allowed cost reductions of more than 50 % because the volumetric ratio of Fe^0 should not exceed 50 %. The establishment of Fe^0 -based technologies can be regarded as the scientific challenge of the next years. In particular, this is because of the appropriateness of Fe^0 -based systems for decentralized water treatment.

7. Epilogue

The presented work corresponds to the original manuscript evaluated by three anonymous referees. Minor revisions were performed, stricktly limited at actualising bibliographic references. One significant concern was raised by one referee: a small summary section supported by a table comparing the costs of the presented Fe^0 -based treatment systems would have enabled sound scientific conclusions. This concern is not addressed here for one major reason: the subject of the thesis is that past Fe^0 -based experiments cannot be compared because of the different set ups used and because of intrinsic differences between used Fe^0 materials used. The fundamental point that tested and used Fe^0 materials are not comparable in their intrinsic reactivity was discussed in this thesis (Section 2.2) using references from Reardon (1995, 2005), Noubactep et al. (2005a, 2009) and Btatkeu et al. (2013). This view is largely confirmed by a recent overview article by (Birke et al. 2015).

Beside intrinsic differences related to the material reactivity, many researchers have tested Fe^{0} -based systems as 'treatement units' in a chain (Gottinger et al. 2010, Gottinger et al. 2013, Kowalski 2014, Kowalski and Søgaard 2014). For example, Kowalski and Søgaard (2014) used Fe^{0} -units to generate iron oxide for As co-precipitation in a separate sand filter. The approach presented herein is that of Fe^{0} being a stand-alone treatment unit possibly coupled to other units in a treatment train (Rahman et al. 2013, Tepong-Tsindé et al. 2015a). Recent publications have demonstrated that the lack of a holistic approach in investigating complex systems is a general concern for environmental remediation (Westholm et al. 2014, Zafarani et al. 2014, Domga et al. 2015, Tepong-Tsindé et al. 2015a). Coming back to Fe^{0} based systems, it is hoped that science-based designed system will soon accelerated the design of efficient systems for environmental remediation, safe drinking water provision and wastewater treatment worldwide.

8. References

- Acero J.L., Benitez F.J., Real F.J., Roldan G. (2010): Kinetics of aqueous chlorination of some pharmaceuticals and their elimination from water matrices. Water Res. 44, 4158–4170.
- Adams C., Asce M., Wang Y., Loftin K., Meyer M. (2002): Removal of antibiotics from surface and distilled water in conventional water treatment processes. J. Environ. Eng. 128, 253–260.
- Adewuyi Y.G. (2001): Sonochemistry: environmental science and engineering applications. Ind. Eng. Chem. Res. 40, 4681–4715.
- Ali I. (2012): New generation adsorbents for water treatment. Chem. Rev. 112, 5073–5091.
- Andreozzi R. Caprio V. Insola A. Marotta R. (1999): Advanced oxidation processes (AOP) for water purification and recovery, Catal. Today 53, 51–59.
- Andreozzi R., Canterino M., Marotta R., Paxeus N. (2005): Antibiotic removal from wastewaters: the ozonation of amoxicillin. J. Hazard. Mater. 122, 243–250.
- Anipsitakis G.P., Dionysiou D.D. (2004): Radical generation by the interaction of transition metals with common oxidants, Environ. Sci. Technol. 38, 3705–3712.
- Antonio M.R., Karet G.B., Guzowski Jr. J.P. (2000): Iron chemistry in petroleum production. Fuel 79, 37–45.
- Antoniou M.G., de la Cruz A.A., Dionysiou D.D. (2010): Degradation of microcystin-LR using sulfate radicals generated through photolysis, thermolysis, and electron transfer mechanisms, Appl. Catal. B- Environ. 96, 290–298.
- Arikan O.A. (2008): Degradation and metabolization of chlortetracycline during the anaerobic digestion of manure from medicated calves. J. Hazard. Mater. 158, 485–490.
- Arnold W.A., Roberts A.L. (1998): Pathways of chlorinated ethylene and chlorinated acetylene reaction with Zn(0). Environ. Sci. Technol. 32, 3017–3025.
- Arnold W.A., Roberts A.L. (2000a): Inter- and Intraspecies Competitive Effects in Reactions of Chlorinated Ethylenes with Zero-Valent Iron in Column Reactors. Environ. Eng. Sci. 17, 291–302.
- Arnold W.A., W.L. Roberts (2000b): Pathways and kinetics of chlorinated ethylene and chlorinated acetylene reaction with Fe(0) particles. Environ. Sci. Technol. 34, 1794–1805.
- Arslan-Alaton I., Gurses F. 2004. Photo-Fenton-like and Fenton-like oxidation of Procaine Penicillin G formulation effluent. J. Photochem. Photobiol. A 165, 165–175.
- Asmus K.-D., (1979): Stabilization of oxidized sulfur centers in organic sulfides. Radical cations and odd-electron sulfur–sulfur bonds. Acc. Chem. Res. 12, 436–442.
- Augugliaro V., Gárcia-López E., Loddo V., Malato-Rodríguez S., Maldonado I., Marcì G., Molinari R., Palmisano L. (2005): Degradation of lincomycin in aqueous medium: coupling of solar photocatalysis and membrane separation. Sol. Energy 79, 402–408.
- Austin P.C. (1911): The interaction of silver nitrate and potassium persulphate and its catalytic effect in the oxidation of organic substances. J. Chem. Soc. Transactions 99, 262–266.
- Ayoub G., Ghauch. A. (2014): Assessment of bimetallic and trimetallic iron-based systems for persulfate activation: application to sulfamethoxazole degradation. Chem. Eng. J. 256, 280–292.

- Bautitz I.R., Nogueira R.F.P. (2010): Photodegradation of lincomycin and diazepam in sewage treatment plant effluent by photo-Fenton process. Catal. Today 151, 94–99.
- Begum A., Gautam S.K. (2011): Dechlorination of endocrine disrupting chemicals using Mg⁰/ZnCl₂ bimetallic system. Water Res. 45 (7), 2383–2391.
- Bigg T., Judd S.J. (2000): Zero-valent iron for water treatment. Environ. Technol. 21, 661–670.
- Bilardi S., Calabrò P.S., Caré S., Moraci N., Noubactep C. (2013a): Improving the sustainability of granular iron/pumice systems for water treatment. J. Environ. Manage. 121, 133–141.
- Bilardi S., Calabrò P.S., Caré S., Moraci N., Noubactep C. (2013b): Effect of pumice and sand on the sustainability of granular iron beds for the removal of Cu^{II}, Ni^{II}, and Zn^{II}. Clean Soil, Air, Water 41, 835–843.
- Birke V., Schuett C., Burmeier H., Friedrich H.-J. (2015): Impact of trace elements and impurities in technical zero-valent iron brands on reductive dechlorination of chlorinated ethenes in groundwater. In Permeable Reactive Barrier Sustainable Groundwater Remediation, R. Naidu and V. Birke (Editors), CRC Press, pp. 87–98, ISBN: 978-1-4822-2447-4.
- Bojic A.Lj., Bojic D., Andjelkovic T. (2009): Removal of Cu²⁺ and Zn²⁺ from model wastewaters by spontaneous reduction–coagulation process in flow conditions. J. Hazard. Mater. 168, 813–819.
- Bokare V., Jung J.-l., Chang Y.-Y., Chang Y.-S. (2013): Reductive dechlorination of octachlorodibenzo-p-dioxin by nanosized zero-valent zinc: Modeling of rate kinetics and congener profile. J. Hazard. Mater. 250-251, 397–402.
- Bonin P.M.L., Odziemkowski M.S., Gillham R.W. (1998): Influence of chlorinated solvents on polarization and corrosion behaviour of iron in borate buffer, Corros. Sci. 40, 1391–1409.
- Boreen A.L., Arnold W.A., McNeill K. (2004): Photochemical fate of sulfa drugs in the aquatic environment: sulfa drugs containing five-membered heterocyclic groups. Environ. Sci. Technol. 38, 3933–3940.
- Boukari S.O.B., Pellizzari F., Karpel Vel Leitner N. (2011): Influence of persulfate ions on the removal of phenol in aqueous solution using electron beam irradiation. J. Hazard. Mater. 185, 844–851.
- Brondel D., Edwards R., Hayman A., Hill D., Mehta S., Semerad T. (1994): Corrosion in the oil industry. Oilfield Rev. 6, 4–18.
- Brown K.D., Kulis J., Thomson B., Chapman T.H., Mawhinney D.B. (2006): Occurrence of antibiotics in hospital, residential, and dairy effluent, municipal wastewater, and the Rio Grande in New Mexico. Sci. Total Environ. 366, 772–783.
- Btatkeu K. B.D., Miyajima K., Noubactep C., Caré S. (2013): Testing the suitability of metallic iron for environmental remediation: Discoloration of methylene blue in column studies. Chem. Eng. J. 215-216, 959–968.
- Burghardt D., Kassahun A. (2005): Development of a reactive zone technology for simultaneous in situ immobilisation of radium and uranium. Environ. Geol. 49, 314–320.

- Buxton G.V., Greenstock C.L. Helman, W.P. Ross, A.B. (1988): Critical review of rate constants for reactions of hydrated electrons, hydrogen atoms and hydroxyl radicals (•OH/•O⁻) in aqueous solution. J. Phys. Chem. Ref. Data 17, 513–531.
- Cai K., Phillips D.H., Elliott C.T., Muller M., Scippo M.-L., Connolly L. (2013): Removal of natural hormones in dairy farm wastewater using reactive and sorptive materials. Sci. Tot. Environ. 461-462, 1–9.
- Calza, P., Medana, C., Pazzi, M., Baiocchi, C., Pelizzetti, E., (2004): Photocatalytic transformations of sulphonamides on titanium dioxide. Appl. Catal. B 53, 63–69.
- Cao M., Wang L., Wang L., Chen J., Lu X. (2013): Remediation of DDTs contaminated soil in a novel Fenton-like system with zero-valent iron, Chemosphere 90, 2303–2308.
- Caré S., Nguyen Q.T., L'Hostis V., Berthaud Y. (2008): Mechanical properties of the rust layer induced by impressed current method in reinforced mortar. Cement Concrete Res. 38, 1079–1091.
- Caré S., Nguyen Q.T., Beddiar K., Berthaud Y. (2010): Times to cracking in reinforced mortar beams subjected to accelerated corrosion tests. Mater. Struct. 43, 107–124.
- Caré S., Crane R., Calabro P.S., Ghauch A., Temgoua E., Noubactep C. (2013): Modelling the permeability loss of metallic iron water filtration systems. Clean Soil, Air, Water 41, 275–282.
- Caule E.J., Cohen M. (1955): The formation of thin films of iron oxide. Can. J. Chem. 33, 288–304.
- Charlet L., Liger E., Gerasimo P. (1998): Decontamination of TCE- and U-rich waters by granular iron: role of sorbed Fe (II). J. Environ. Eng. 124, 25–30.
- Chen L., Jin S., Fallgren P.H., Swoboda-Colberg N.G., Liu F., Colberg P.J.S. (2012a): Electrochemical depassivation of zero-valent iron for trichloroethene reduction. J. Hazard. Mater. 239-240, 265–269.
- Chen, X., Xue, Z., Yao, Y., Wang, W., Zhu, F., Hong, C. (2012b): Oxidation degradation of rhodamine B in aqueous by UV/S₂O₈²⁻ treatment system. Int. J. Photoenergy, art. no. 754691.
- Chen L., Jin S., Fallgren P.H., Liu F., Colberg P.J.S. (2013): Passivation of zero-valent iron by denitrifying bacteria and the impact on trichloroethene reduction in groundwater. Water Sci. Technol. 67, 1254–1259.
- Chiu P.C. (2013): Applications of zero-valent iron (ZVI) and nanoscale ZVI to municipal and decentralized drinking water systems – A Review. In Novel Solutions to Water Pollution, Ahuja S. and Hristovski K. (Eds), ACS Symposium Series, Vol. 1123; American Chemical Society: Washington, DC, doi: 10.1021/bk-2013-1123.ch014, pp 237–249.
- Choi K.-J., Son H.-J., Kim S.-H. (2007): Ionic treatment for removal of sulphonamide and tetracycline classes of antibiotic. Sci. Total Environ. 387, 247–256.
- Cokgor E.U., Arslan-Alaton I., Karahan O., Dogruel S., Orhon D. (2004): Biological treatability of raw and ozonated penicillin formulation effluent. J. Hazard. Mater. B116, 159–166.
- Comba S., Di Molfetta A., Sethi R. 2011: A comparison between field applications of nano-, micro-, and millimetric zero-valent iron for the remediation of contaminated aquifers. Water, Air, Soil Poll. 215, 595–607.

- Corma A. Garcia H. Chem. Rev. (2003): Lewis Acids: From conventional homogeneous to green homogeneous and heterogeneous catalysis. 103, 4307–4366.
- Correia de Velosa A., Pupo Nogueira R.F. (2013): 2,4-Dichlorophenoxyacetic acid (2,4-D) degradation promoted by nanoparticulate zerovalent iron (nZVI) in aerobic suspensions. J. Environ. Manage. 121, 72–79.
- Crane R.A., Scott T.B. 2012: Nanoscale zero-valent iron: Future prospects for an emerging water treatment technology. J. Hazard. Mater. 211-212, 112–125.
- Crane R., Noubactep C. (2012): Elemental metals for environmental remediation: learning from hydrometallurgy. Fresenius Environ. Bull. 21, 1192–1196.
- Criquet J., Karpel Vel Leitner N. (2011): Electron beam irradiation of aqueous solution of persulfate ions. Chem. Eng. J. 169, 258–262.
- Crisafully R., Milhome M.A.L., Cavalcante R.M., Silveira E.R., De Keukeleire D., Nascimento R.F., (2008): Removal of some polycyclic aromatic hydrocarbons from petrochemical wastewater using low-cost adsorbents of natural origin. Bioresour. Technol. 99, 4515–4519.
- Daigger G.T. (2003): Tools for future success. Water Environ. Technol. 15, 38-45.
- Daughton C.G., Ternes T.A. (1999): Pharmaceuticals and personal care products in the environment: Agents of subtle change? Environ. Health Persp. 107, 907–938.
- Daughton C.G. (2002): Environmental stewardship and drugs as pollutants, *Lancet* 360, 1035-1036, 2002.
- Daughton C.G. (2007): Pharmaceuticals in the environment: sources and their management, Comprehensive Analytical Chemistry 50, Petrovic and Barcelo (Eds) Chapter 1, 1–69.
- David B. (2009): Sonochemical degradation of PAH in aqueous solution. Part I: monocomponent PAH solution. Ultrason. Sonochem. 16, 260–265.
- De Laat J., Le T.G. (2005): Kinetics and modeling of the Fe(III)/H₂O₂ system in the presence of sulfate in acidic aqueous solutions. Environ. Sci. Technol. 39, 1811–1818.
- Deng Y., Ezyske C.M. (2011): Sulfate radical-advanced oxidation process (SR-AOP) for simultaneous removal of refractory organic contaminants and ammonia in landfill leachate. Water Res. 45, 6189–6194.
- Dhakshinamoorthy A., Navalon S., Alvaro. M., Garcia H., (2012): Metal nanoparticles as heterogeneous Fenton catalysts, ChemSusChem. (5) 46–64.
- Dickerson R.E., Gray H.B., Haight Jr., G.P. (1979): Chemical Principles, 3. Edition, Benjamin/Cummings Inc. London, Amsterdam, 944 pp.
- Diya'uddeen B.H., Aziz A.R.A., Daud W.M.A.W. (2012) Oxidative mineralization of petroleum refinery effluent using Fenton-like process. Chem. Eng. Res. Des. 90, 298–307.
- Domga R., Togue-Kamga F., Noubactep C., Tchatchueng J.-B. (2015): Discussing porosity loss of Fe⁰ packed water filters at ground level. Chem. Eng. J. 263, 127–134.
- Droste R.L. (1997): Theory and Practice of Water and Wastewater Treatment, John Wiley & Sons, Inc., New York, 1997.
- Du J., Liu Y., Zhang X., Qiu M., Zhang L., Fan Y. (2013): Fe⁰/Zn⁰ PRB technology for the remediation of PCBs contaminated groundwater. Appl. Mechanics Mater. 295-298, 1364–1367.
- Dyer E., Pickett Jr O.A., Strause S.F., Worrell Jr H.E. (1956) The persulfate-initiated oxidation of vinyl monomers in aqueous solutions. J. Am. Chem. Soc. 78, 3384–3388.

- Dyson T. (1996): Population and food: global trends and future prospects, Routledge. London.
- Elliot W.D., Zhang W-X. (2001): Field assessment of nanoscale bimetallic particles for groundwater treatment. 35, 4922–4926.
- Elsner M., Cwiertny D.M., Roberts A.L., Lollar B.S. (2007): Response to comment on "1,1,2,2-Tetrachloroethane reactions with OH-, Cr(II), granular iron, and a copper-iron bimetal: Insights from product formation and associated carbon isotope fractionation". Environ. Sci. Technol. 41, 7949–7950.
- Eusterhues K., Rennert T., Knicker H., Kgel-Knabner I., Totsche K.U., Schwertmann U. (2011): Fractionation of organic matter due to reaction with ferrihydrite: Coprecipitation versus adsorption. Environ. Sci. Technol. 45, 527–533.
- Fang G., Gao, J., Dionysiou D.D., Liu C., Zhou D. (2013a): Activation of persulfate by quinones: Free radical reactions and implication for the degradation of PCBs. Environ. Sci. Technol. 47, 4605–4611.
- Fang G.-D., Dionysiou D.D., Zhou D.-M., Wang Y., Zhu X.-D., Fan J.-X., Cang L., Wang Y.-J. (2013b): Transformation of polychlorinated biphenyls by persulfate at ambient temperature, Chemosphere 90, 1573–1580.
- Fang J.-Y., Shang C. (2012): Bromate formation from bromide oxidation by the UV/persulfate process. Environ. Sci. Technol. 46, 8976–8983.
- Feitosa-Felizzola J., Chiron S. (2009): Occurrence and distribution of selected antibiotics in a small Mediterranean stream (Arc River, Southern France). J. Hydrol. 364, 50–57.
- Fenton H.J.H. (1894): Oxidation of tartaric acid in presence of iron. J. Chem. Soc. 65, 899–910.
- Forsey S. P. (2004): In situ chemical oxidation of creosote/coal tar residuals: Experimental and numerical investigation. Earth Sciences. Waterloo, University of Waterloo. Ph.D.
- Franklin L.B. (1991): Wastewater Engineering: Treatment. Disposal and Reuse, McGraw Hill, Inc., New York.
- Frost R.L., Xi Y., He H. (2010): Synthesis, characterization of palygorskite supported zerovalent iron and its application for methylene blue adsorption. J. Colloid Interf. Sci. 341, 153–161.
- Furukawa Y., Kim J.-W., Watkins J., Wilkin R.T. (2002): Formation of ferrihydrite and associated iron corrosion products in permeable reactive barriers of zero-valent iron. Environ. Sci. Technol. 36, 5469–5475.
- Gan S., Lau E.V., Ng H.K. (2009): Remediation of soils contaminated with polycyclic aromatic hydrocarbons (PAHs). J. Hazard. Mater. 172, 532–549.
- Gao, Y.-Q., Gao, N.-Y., Deng, Y., Yang, Y.-Q., Ma, Y. (2012): Ultraviolet (UV) lightactivated persulfate oxidation of sulfamethazine in water. Chem. Eng. J. 195-196, 248– 253.
- Ghauch A. (2008a): Rapid removal of flutriafol from water by zero valent iron powder. Chemosphere 71, 816–826.
- Ghauch A. (2008b): Discussion of Chicgoua Noubactep on "Removal of thiobencarb in aqueous solution by zero valent iron" by Md. Nurul Amin et al. [Chemosphere 70 (2008) 511-515]. Chemosphere 72, 328-331.

- Ghauch A., Tuqan A. (2008): Catalytic degradation of chlorothalonil in water using bimetallic iron-based systems. Chemosphere 73, 751–759.
- Ghauch A., Tuqan A. (2009): Reductive destruction and decontamination of aqueous solutions of chlorinated antimicrobial agent using bimetallic systems. J. Hazard. Mater. 169, 665–674.
- Ghauch A., Tuqan A. Abou Assi H. (2009): Antibiotic removal from water: Elimination of amoxicillin and ampicillin by microscale and nanoscale iron particles. Environ. Pollut. 157, 1626–1635.
- Ghauch A. (2010): Response to Dr. Franska's comments on the paper "Antibiotic removal from water: Elimination of amoxicillin and ampicillin by microscale and nanoscale iron particles." Ghauch et al. (2009) Environ. Pollut. 157, 1626-1635. Environ. Pollut. 158, 3030–3031.
- Ghauch A., Abou Assi H., Tuqan A. (2010a): Investigating the mechanism of clofibric acid removal in Fe⁰/H₂O systems. J. Hazard. Mater. 176, 48–55.
- Ghauch A., Abou Assi H., Bdeir S. (2010b): Aqueous removal of diclofenac by plated elemental iron: bimetallic systems, J. Hazard. Mater. 182, 64–74.
- Ghauch A., Abou Assi H., Baydoun H., Tuqan A., Bejjani A. (2011a): Fe⁰-based trimetallic systems for the removal of aqueous diclofenac: mechanism and kinetics, Chem. Eng. J. 172, 1033–1044.
- Ghauch A., Baydoun H., Tuqan A., Ayoub G., Naim, S. (2011b): Submicrometric iron particles for the removal of pharmaceuticals in water: Application to b-lactam antibiotics. Adv. Mater. Res. 324, 485–488.
- Ghauch A., Baydoun H., Dermesropian, P. (2011c): Degradation of aqueous carbamazepine in ultrasonic/Fe⁰/H₂O₂ systems, Chem. Eng. J. 172, 18–27.
- Ghauch A., Tuqan A.M. (2012): Oxidation of bisoprolol in heated persulfate/H2O systems: kinetics and products, Chem. Eng. J. 183, 162–171.
- Ghauch A., Tuqan A.M., Kibbi N. (2012): Ibuprofen removal by heated persulfate in aqueous solution: a kinetics study. Chem. Eng. J. 197, 483–492.
- Ghauch A., Tuqan A.M., Kibbi N. Geryes S. (2012): Methylene Blue discoloration by heated persulfate in aqueous solution. Chem. Eng. J. 213, 259–271.
- Ghauch A., Ayoub G. Naim S. (2013): Degradation of sulfamethoxazole by persulfate assisted micrometric Fe0 in aqueous solution. Chem. Eng. J. 228, 1168–1181.
- Gheju M. (2011): Hexavalent chromium reduction with zero-valent iron (ZVI) in aquatic systems. Water, Air, Soil Poll. 222, 103–148.
- Gheju M., Balcu I. (2011): Removal of chromium from Cr(VI) polluted wastewaters by reduction with scrap iron and subsequent precipitation of resulted cations. J. Hazard. Mater. 196, 131–138.
- Giles D.E., Mohapatra M., Issa T.B., Anand S., Singh P. (2011): Iron and aluminium based adsorption strategies for removing arsenic from water. J. Environ. Manage. 92, 3011–3022.
- Gillham R.W., O'Hannesin S.F. (1994): Enhanced degradation of halogenated aliphatics by zerovalent iron. Ground water 32, 958–967.

- Goel M., Hongqiang H., Mujumdar A.S., Ray M.B. (2004): Sonochemical decomposition of volatile and non-volatile organic compounds a comparative study. Water Res. 38, 4247–4261.
- González O., Sans C., Esplugas S. (2007): Sulfamethoxazole abatement by photo-Fenton. Toxicity, inhibition and biodegradability assessment of intermediates. J. Hazard. Mater. 146, 459–464.
- Gosetti F., Gianotti V., Angioi S., Polati S., Marengo E., Gennaro M.C. (2004): Oxidative degradation of food dye E133 Brilliant Blue FCF: Liquid chromatography–electrospray mass spectrometry identification of the degradation pathway. J. Chromat. A 1054, 379–387.
- Gottinger A.M., Wild D.J., McMartin D., Moldovan B., Wang. D. (2010): Development of an iron-amended biofilter for removal of arsenic from rural Canadian prairie potable water.In: Water Pollution X. A.M. Marinov and C.A. Brebbia, Eds.; WIT Press: Ashurst, Southampton, 2010, 333–344.
- Gottinger A.M., McMartin D.W., Wild D.J., Moldovan B. (2013): Integration of zero valent iron sand beds into biological treatment systems for uranium removal from drinking water wells in rural Canada. Can. J. Civ. Eng. 40, 945–950.
- Gould J.P. (1982): The kinetics of hexavalent chromium reduction by metallic iron. Water Res. 16 (6), 871–877.
- Grittini C., Malcomson M., Fernando Q., Korte N. (1995): Rapid dechlorination of polychlorinated biphenils on the surface of a Pd/Fe bimetallic system. Environ. Sci. Technol. 29, 2898–2900.
- Gu B., Schmitt J., Chen Z., Liang L., McCarthy J.F. (1994): Adsorption and desorption of natural organic matter on iron oxide: mechanisms, and models. Environ. Sci. Technol. 28, 38–46.
- Haag W.R., Yao C.C. (1992): Rate constants for reaction of hydroxyl radicals with several drinking-water contaminants. Environ. Sci. Technol. 26, 1005–1013.
- Hao C., Clement R., Yang P. (2007): Liquid chromatography tandem mass spectrometry of bioactive pharmaceutical compounds in the aquatic environmentda decade's activities. Anal. Bioanal. Chem. 378, 1247–1257.
- He X., de la Cruz A.A., Dionysiou D.D. (2013): Destruction of cyanobacterial toxin cylindrospermopsin by hydroxyl radicals and sulfate radicals using UV-254nm activation of hydrogen peroxide, persulfate and peroxymonosulfate. J. Photochem. Photobiol. A: Chemistry 251, 160–166.
- Heberer T. (2002). Occurrence, fate, and removal of pharmaceutical residues in the aquatic environment: A review of recent research data. Toxicol. Lett. 131, 5–17.
- Henderson A.D., Demond A.H. (2007): Long-term performance of zero-valent iron permeable reactive barriers: a critical review. Environ. Eng. Sci. 24, 401–423.
- Henderson A.D., Demond A.H. (2011): Impact of solids formation and gas production on the permeability of ZVI PRBs. J. Environ. Eng. 137, 689–696.
- Henderson A.D., Demond A.H. (2013): Permeability of iron sulfide (FeS)-based materials for groundwater remediation. Water Res. 47, 1267–1276.
- Hendriks A.J. (2013): How to deal with 100,000+ substances, sites, and species: Overarching principles in environmental risk assessment. Environ. Sci. Technol. 47, 3546–3547.

- Hernando M.D., Mezcua M., Fernandez-Alba A.R., Barcelo D. (2006): Environmental risk assessment of pharmaceutical residues in wastewater effluents, surface waters and sediments, Talanta 69, 334–342.
- Hillebrand O., Musallam S., Scherer L., Nödler K., Licha T. (2013): The challenge of samplestabilisation in the era of multi-residue analytical methods: A practical guideline for the stabilisation of 46 organic micropollutants in aqueous samples. Science of the Total Environment 454–455, 289–298.
- Hirose J., Kondo F., Nakano T., Kobayashi T., Hiro N., Ando Y., Takenaka H., Sano K. (2005): Inactivation of antineoplastics in clinical wastewater by electrolysis. Chemosphere 60, 1018–1024.
- Hirsch R., Ternes T., Haberer K., Kratz K.-L. (1999): Occurrence of antibiotics in the aquatic environment. Sci. Total Environ. 225, 109–118.
- House D.A. (1962): Kinetics and mechanisms of oxidations by peroxidisulfate. Chem. Rev. 62, 185–203.
- Hu W., Li S.-X., Wang C.-D. (2013): Decolourization of methylene blue by persulfate. Adv. Mater. Res. 634-638, 1, 76–80.
- Huang K.C. Zhao Z. Hoag, G.E. Dahmani A. Block P.A. (2005): Degradation of volatile organic compounds with thermally activated persulfate oxidation, Chemosphere 61, 551–560.
- Huang Y.H., Zhang T.C. (2005): Enhancement of nitrate reduction in Fe⁰-packed columns by selected cations. J. Environ. Eng. 131, 603–611.
- Huang P., Ye Z., Xie W., Chen Q., Li J., Xu Z., Yao M. (2013a): Rapid magnetic removal of aqueous heavy metals and their relevant mechanisms using nanoscale zero valent iron (nZVI) particles. Water Res. 47, 4050–4058.
- Huang W., Liu F., Yan N., Lu A., Chen H., Chen L., Wang F. (2013b): Removal of trichloroethylene in groundwater with two oxidants: Siderite catalyzed hydrogen peroxide and sodium persulfate. Wat. Sci. Technol. 13, 36–43.
- Huie R.E., Clifton C.L., Neta P. (1991): Electron transfer reaction rates and equilibria of the carbonate and sulfate radical anions. Radiat. Phys. Chem. 38, 477–481.
- Huling S.G., Ko S., Park S., Kan E. (2011): Persulfate oxidation of MTBE- and chloroformspent granular activated carbon, J. Hazard. Mater. 192, 1484–1490.
- Hussain I., Zhang Y., Huang S., Du X. (2012): Degradation of p-chloroaniline by persulfate activated with zero-valent iron, Chem. Eng. J. 203, 269–276.
- Hussam A., Munir A.K.M. (2007): A simple and effective arsenic filter based on composite iron matrix: Development and deployment studies for groundwater of Bangladesh. J. Environ. Sci. Health A 42, 1869–1878.
- Hussam A. (2009): Contending with a Development Disaster: SONO Filters Remove Arsenic from Well Water in Bangladesh. Innovations 4, 89–102.
- Hussam A. (2013): Potable water: Nature and purification. Monitoring Water Quality 261–283.
- ITRC (2011): (Interstate Technology & Regulatory Council) Permeable Reactive Barrier: Technology Update. PRB-5. Washington, D.C.: Interstate Technology & Regulatory Council, PRB: Technology Update Team. <u>www.itrcweb.org</u>.

- Jambor J.L. Raudsepp M., Mountjoy K. (2005): Mineralogy of permeable reactive barriers for the attenuation of subsurface contaminants. Can. Miner. 43, 2117–2140.
- Jara C.C., Fino D., Specchia V., Saracco G., Spinelli P. (2007): Electrochemical removal of antibiotics from wastewater. Appl. Catal. B 70, 479–487.
- Jia Y., Aagaard P., Breedveld G.D. (2007): Sorption of triazoles to soil and iron minerals. Chemosphere 67, 250–258.
- Jiao S., Zheng S., Yin D., Wang L., Chen, L. (2008): Aqueous photolysis of tetracycline and toxicity of photocatalytic products to luminescent bacteria. Chemosphere 73, 377–382.
- Jiao Y., Qiu C., Huang L., Wu K., Ma H., Chen S., Ma L., Wu L. (2009): Reductive dechlorination of carbon tetrachloride by zero-valent iron and related iron corrosion. Appl. Catal. B: Environ. 91, 434–440.
- Kang N., Hua I. (2005): Enhanced chemical oxidation of aromatic hydrocarbons in soil systems. Chemosphere 61, 909–922.
- Kaplan S. (2013): Pharmacological pollution in water. Crit. Rev. Environ. Sci. Technol. 43, 1074–1116.
- Kim S.H., Shon H.K., Ngo H.H. (2010): Adsorption characteristics of antibiotics trimethoprim on powered and granular activated carbon. J. Ind. Eng. Chem. 16, 344–349.
- Kishimoto N., Iwano S., Narazaki Y. (2011): Mechanistic consideration of zinc ion removal by zero-valent iron. Water Air Soil Pollut. 221, 183–189.
- Klauson D., Babkina J., Stepanova K., Krichevskaya M., Preis S., (2010): Aqueous photocatalytic oxidation of amoxicillin. Catal. Today, 151, 39–45.
- Klavarioti M., Mantzavinos D., Kassinos D. (2009): Removal of residual pharmaceuticals from aqueous systems by advanced oxidation processes. Environ. Int. 35, 402–417.
- Klosterhaus S.L., Grace R., Hamilton M.C., Yee D. (2013): Method validation and reconnaissance of pharmaceuticals, personal care products, and alkylphenols in surface waters, sediments, and mussels in an urban estuary. Environ. Int. 54, 92–99.
- Kolthoff I.M., Medalia A.I., Raaen H.P. (1951): The Reaction between Ferrous Iron and Peroxides. IV. Reaction with Potassium Persulfate. J. Am. Chem. Soc. 73, 1733–1739.
- Kolthoff I.M., Miller I.K. (1951): The chemistry of persulfate. I. The kinetics and mechanism of the decomposition of the persulfate ion in aqueous medium. J. Am. Chem. Soc., 73, 3055–3059.
- Kosutic K., Dolar D., Asperger D., Kunst B. (2007): Removal of antibiotic from model wastewater by RO/NF membrane. Sep. Purif. Technol. 53, 244–249.
- Kowalski K. (2014): New method for arsenic compounds elimination from naturally contaminated drinking water systems. Ph.D. Thesis, Aalborg Universitet, Esbjerg, Denmark.
- Kowalski K.P., Søgaard E.G. (2014): Implementation of zero-valent iron (ZVI) into drinking water supply Role of the ZVI and biological processes. Chemosphere 117, 108–114.
- Koyuncu I., Arikan O.A., Wiesner M.R., Rice C. (2008): Removal of hormones and antibiotics by nanofiltration membranes. J. Membr. Sci. 309, 94–101.
- Ku Y., Tu Y.H., Ma C.M. (2005): Effect of hydrogen peroxide on the decomposition of monochlorophenols by sonolysis in aqueous solution. Water Res. 39, 1093–1098.
- Kümmerer K. (2009): The presence of pharmaceuticals in the environment due to human use present knowledge and future challenges. J. Environ. Manage. 90, 2354–2366.

- Lackovic J.A., Nikolaidis N.P., Dobbs G.M. (2000): Inorganic arsenic removal by zero-valent iron. Environ. Eng. Sci. 17, 29–39.
- Lai B., Chen Z., Zhou Y., Yang P., Wang J., Chen Z. (2013): Removal of high concentration p-nitrophenol in aqueous solution by zero valent iron with ultrasonic irradiation (US-ZVI). J. Hazard. Mater. 250-251, 220–228.
- Lai K.C.K., Lo I.M.C., Birkelund V., Kjeldsen P. (2006): Field monitoring of a permeable reactive barrier for removal of chlorinated organics. J. Environ. Eng. 132, 199–210.
- Lavine B.K., Auslander G., Ritter J. (2001): Polarographic studies of zero valent iron as a reductant for remediation of nitroaromatics in the environment. Microchem. J., 70, 69– 83.
- Lewis A. (2010): Review of metal sulphide precipitation. Hydrometallurgy 104, 222–234.
- Lewis A., Swartbooi A. (2006): Factors affecting metal removal in mixed sulfide precipitation. Chem. Eng. Technol. 29, 277–280.
- Lewis A., van Hille R. (2006): An exploration into the sulphide precipitation method and its effect on metal sulphide removal. Hydrometallurgy 81, 197–204.
- Li B., Li L, Lin K, Zhang W, Lu S, Luo Q, (2009): Removal of 1,1,1-trichloroethane from aqueous solution by a sono-activated persulfate process. Ultrason. Sonochem. 20, 855–863.
- Li L., Van Genuchten C.M., Addy S.E.A., Yao J., Gao N., Gadgil A.J. (2012): Modeling As(III) oxidation and removal with iron electrocoagulation in groundwater, Environ. Sci. Technol. 46, 12038–12045.
- Li S.-Z., Li X.-Y., Wang D.-Z. (2004): Membrane (RO-UF) filtration for antibiotic wastewater treatment and recovery of antibiotics. Sep. Purif. Technol. 34, 109–114.
- Li T., Chen Y., Wan P., Fan M., Yang X.J. (2010): Chemical degradation of drinking water disinfection byproducts by millimeter-sized particles of iron-silicon and magnesium-aluminum alloys. J. Am. Chem. Soc. 132, 2500–2501.
- Liang L., Korte N., Goodlaxson J.D., Clausen J., Fernando Q., Muftikian R. (1997): Byproduct formation during the reduction of TCE by zero-valence iron and palladized iron. Ground Water Monit. Remed. 17, 122–127.
- Liang C. Bruell C.J. Marley M.C. Sperry K.L. (2003): Thermally activated persulfate oxidation of trichloroethylene (TCE) and 1,1,1-trichloroethane (TCA) in aqueous systems and soil slurries. Soil Sediment Contam. 12, 207–228.
- Liang C. Bruell C.J. Marley M.C. Sperry K.L. (2004): Persulfate oxidation for in situ remediation of TCE. II. Activated by chelated ferrous ion. Chemosphere 55, 1225–1233.
- Liang C., Huang C.F., Chen Y.J. (2008): Potential for activated persulfate degradation of BTEX contamination. Water Res. 42, 4091–4100.
- Lien H.-L., Zhang W. (2002): Enhanced dehalogenation of halogenated methanes by bimetallic Cu/Al. Chemosphere 49, 371–378.
- Lin Y.T., Liang C., Chen J.H. (2011a): Feasibility study of ultraviolet activated persulfate oxidation of phenol. Chemosphere 82, 1168–1172.
- Lin Y.-F., Chen C.-H., Xie W.-J., Yang S.-H., Hsu C.-S., Lin M.-T., Jian W.-B. (2011b): Nano approach investigation of the conduction mechanism in polyaniline nanofibers. ACS Nano 5, 1541–1548.

- Lipczynska-Kochany E., Harms S., Milburn R., Sprah G., Nadarajah N. (1994): Degradation of carbon tetrachloride in the presence of iron and sulphur containing compounds. Chemosphere 29, 1477–1489.
- Lipczynska-Kochany E., Sprah G., Harms S. (1995): Influence of some groundwater and surface waters constituents on the degradation of 4-chlorophenol by the Fenton reaction. Chemosphere 30, 9–20.
- Litter M.I., Morgada M.E., Bundschuh J. (2010): Possible treatments for arsenic removal in Latin American waters for human consumption. Environ. Pollut. 158, 1105–1118.
- Liu C.S., Shih K., Sun C.X., Wang F. (2012): Oxidative degradation of propachlor by ferrous and copper ion activated persulfate. Sci. Total Environ. 416, 507–512.
- Liu H., Wang Q., Wang C., Li X.-z. (2013): Electron efficiency of zero-valent iron for groundwater remediation and wastewater treatment. Chem. Eng. J. 215-216, 90–95.
- Ma L.M., Ding Z.G., Gao T.Y., Zhou R.F., Xu W.Y., Liu J. (2004): Discoloration of methylene blue and wastewater from a plant by a Fe/Cu bimetallic system. Chemosphere 55, 1207–1212.
- Mantha R., Taylor K.E., Biswas N., Bewtra J.K. (2001): A continuous system for Fe⁰ reduction of nitrobenzene in synthetic wastewater. Environ. Sci. Technol. 35, 3231–3236.
- Martínez J.L. (2009): Environmental pollution by antibiotics and by antibiotic resistance determinants. Environ. Pollut 157, 2893–2902.
- Matheson L.J. and Tratnyek P.G. (1994): Reductive dehalogenation of chlorinated methanes by iron metal. Environ. Sci. Technol. 28, 2045–2053.
- Merouani S., Hamdaoui, O., Saoudi F., Chiha M. (2010): Sonochemical degradation of Rhodamine B in aqueous phase: effects of additives. Chem. Eng. J. 158, 550–557.
- Michen B., Meder F., Rust A., Fritsch J., Aneziris C., Graule T. (2012): Virus removal in ceramic depth filters based on diatomaceous earth. Environ. Sci. Technol. 46, 1170–1177.
- Miehr R., Tratnyek G. P., Bandstra Z.J., Scherer M. M., Alowitz J.M., Bylaska J. E. (2004): Diversity of contaminant reduction reactions by zerovalent iron: Role of the reductate. Environ. Sci. Technol. 38, 139–147.
- Minisci F. Citterio A. (1983): Electron-transfer processes: peroxydisulfate, a useful and versatile reagent in organic chemistry. Acc. Chem. Res. 16, 27–32.
- Mitchell G., Poole P., Segrove H.D. (1955): Adsorption of methylene blue by high-silica sands. Nature 176, 1025–1026.
- Miyajima K. (2012): Optimizing the design of metallic iron filters for water treatment. Freiberg Online Geoscience 32, 60 pp. (www.geo.tu-freiberg.de/fog)
- Miyajima K., Noubactep C. (2012): Effects of mixing granular iron with sand on the efficiency of methylene blue discoloration. Chem. Eng. J. 200-202, 433–438.
- Miyajima K., Noubactep C. (2013): Impact of Fe⁰ amendment on methylene blue discoloration by sand columns. Chem. Eng. J. 217, 310–319.
- Miyajima K., Noubactep C. (2015): Characterizing the impact of sand addition on the efficiency of granular iron for water treatment. Chem. Eng. J. 262, 891–896.

- Momba M.N.B., Obi C.L., Thompson P. (2009): Survey of disinfection efficiency of small drinking water treatment plants: Challenges facing small water treatment plants in South Africa. Water SA 35, 485–494.
- Moreno H.A.C., Cocke D.L., Gomes J.A.G., Morkovsky P., Parga J.R., Peterson E., Garcia C. (2009): Electrochemical reactions for electrocoagulation using iron electrodes, Ind. Eng. Chem. Res. 48, 2275–2282.
- Morrison S.J., Mushovic P.S., Niesen P.L. (2006): Early breakthrough of molybdenum and uranium in a permeable reactive barrier. Environ. Sci. Technol. 40, 2018–2024.
- Nagpal V., Bokare A.D., Chikate R.C., Rode C.V., Paknikar K.M. (2012): Reply to comment on "Reductive dechlorination of γ-hexachlorocyclohexane using Fe–Pd bimetallic nanoparticles", by C. Noubactep. J. Hazard. Mater. 235-236, 392–393.
- Nalli K. (2010): Corrosion and its mitigation in the Oil & Gas Industry An overview. PetroMin Pipeliner (Jan-Mar 2010) 10–13.
- Nam S.N., Han S.K., Kang J.W., Choi H. (2003): Kinetics and mechanisms of the sonolytic destruction of non-volatile organic compounds: investigation of the sonochemical reaction zone using several OH monitoring techniques. Ultrason. Sonochem. 10, 139–147.
- Neppolian B., Doronila A., Ashokkumar M. (2010): Sonochemical oxidation of arsenic(III) to arsenic(V) using potassium peroxydisulfate as an oxidizing agent. Water Res. 44, 3687–3695.
- Nesic S. (2007): Key issues related to modelling of internal corrosion of oil and gas pipelines A review. Corros. Sci. 49, 4308–4338.
- Neta P., Madhavan V., Zemel H. (1977): Rate constants and mechanism of reaction of SO₄⁻⁻ with aromatic-compounds. J. Am. Chem. Soc. 99, 163–164.
- Neumann A., Kaegi R., Voegelin A., Hussam A., Munir A.K.M., Hug S.J. (2013): Arsenic removal with composite iron matrix filters in Bangladesh: A field and laboratory study. Environ. Sci. Technol. 47, 4544–4554.
- Nfodzo P., Choi H. (2011): Sulfate radicals destroy pharmaceuticals and personal care products. Environ. Eng. Sci. 28, 605–609.
- Niedbala A., Schaffer M., Licha T., Nödler K., Börnick H., Ruppert H., Worch E. (2013): Influence of competing inorganic cations on the ion exchange equilibrium of the monovalent organic cation metoprolol on natural sediment. Chemosphere 90, 1945– 1951.
- Nödler K., Licha T., Bester K., Sauter M. (2010): Development of a multi-residue analytical method, based on liquid chromatography-tandem mass spectrometry, for the simultaneous determination of 46 micro-contaminants in aqueous samples. J. Chromatogr. A 1217, 6511–6521.
- Noubactep C., Meinrath G., Dietrich P., Sauter M., Merkel B. (2005a): Testing the suitability of zerovalent iron materials for reactive walls. Environ. Chem. 2, 71–76.
- Noubactep C., Meinrath G., Merkel J.B. (2005b): Investigating the mechanism of uranium removal by zerovalent iron materials. Environ. Chem. 2, 235–242.
- Noubactep C. (2006a): Contaminant reduction at the surface of elemental iron: the end of a myth. Wissenschaftliche Mitteilungen, 31: 173–179, Freiberg (ISSN 1433-128).
- Noubactep C. (2006b): Contaminant reduction at the surface of elemental iron: the end of a myth (in German). wlb Wasser, Luft und Boden 50 (11/12), TerraTech 11-12, TT11–14.

- Noubactep C., Schöner A., Meinrath G. (2006): Mechanism of uranium (VI) fixation by elemental iron. J. Hazard. Mater. 132, 202–212.
- Noubactep C. (2007): Processes of contaminant removal in "Fe⁰–H₂O" systems revisited. The importance of co-precipitation. Open Environ. J. 1, 9–13.
- Noubactep C. (2008): A critical review on the mechanism of contaminant removal in Fe^0 – H_2O systems. Environ. Technol. 29, 909–920.
- Noubactep C. (2009b): An analysis of the evolution of reactive species in Fe^0/H_2O systems. J. Hazard. Mater. 168, 1626–1631.
- Noubactep C. (2009a): Characterizing the discoloration of methylene blue in Fe⁰/H₂O systems. J. Hazard. Mater. 166, 79–87.
- Noubactep C. (2009c): Metallic iron for environmental remediation: Learning from the Becher Process. J. Hazard. Mater. 168, 1609–1612.
- Noubactep C. (2009d): On the operating mode of bimetallic systems for environmental remediation. J. Hazard. Mater. 164, 394–395.
- Noubactep C., Licha T., Scott T.B., Fall M., Sauter M. (2009): Exploring the influence of operational parameters on the reactivity of elemental iron materials. J. Hazard. Mater. 172, 943–951.
- Noubactep C. (2010a): The suitability of metallic iron for environmental remediation. Environ. Progr. Sust. En. 29, 286–291.
- Noubactep C. (2010b): The fundamental mechanism of aqueous contaminant removal by metallic iron. Water SA 36, 663–670.
- Noubactep C. (2010c): Elemental metals for environmental remediation: Learning from cementation process. J. Hazard. Mater. 181, 1170–1174.
- Noubactep C., Caré S. (2010a): Enhancing sustainability of household water filters by mixing metallic iron with porous materials. Chem. Eng. J. 162, 635–642.
- Noubactep C., Caré S. (2010b): On nanoscale metallic iron for groundwater remediation. J. Hazard. Mater. 182, 923–927.
- Noubactep C., Caré S., Togue-Kamga F., Schöner A., Woafo P. (2010): Extending service life of household water filters by mixing metallic iron with sand. Clean Soil, Air, Water 38. 951–959.
- Noubactep C., Schöner A. (2010): Metallic iron for environmental remediation: Learning from electrocoagulation. J. Hazard. Mater. 175, 1075–1080.
- Noubactep C. (2011a): Metallic iron for safe drinking water production. Freiberg Online Geology, vol. 27, 38 pp, ISSN 1434-7512. (www.geo.tu-freiberg.de/fog)
- Noubactep C. (2011b): Aqueous contaminant removal by metallic iron: Is the paradigm shifting? Water SA 37, 419–42.
- Noubactep C. (2011c): Metallic iron for water treatment: A knowledge system challenges mainstream science. Fresenius Environ. Bull. 20, 2632–2637.
- Noubactep C. (2011d): On the mechanism microbe inactivation by metallic iron. J. Hazard. Mater. 198, 383–386.
- Noubactep C., Caré S. (2011): Designing laboratory metallic iron columns for better result comparability. J. Hazard. Mater. 189, 809–813.
- Noubactep C. (2012): Investigating the processes of contaminant removal in $Fe^{0}/H_{2}O$ systems. Korean J. Chem. Eng. 29, 1050–1056.
- Noubactep C., Caré S., Btatkeu K.B.D., Nanseu-Njiki C.P. (2012a): Enhancing the sustainability of household Fe⁰/sand filters by using bimetallics and MnO₂. Clean Soil, Air, Water 40, 100–109.
- Noubactep C., Caré S., Crane R.A. (2012b): Nanoscale metallic iron for environmental remediation: prospects and limitations. Water Air Soil Pollut. 223, 1363–1382.
- Noubactep C., Temgoua E., Rahman M.A. (2012c): Designing iron-amended biosand filters for decentralized safe drinking water provision. Clean Soil, Air, Water 40, 798–807.
- Noubactep C. (2013a): Relevant reducing agents in remediation Fe^0/H_2O systems. Clean: Soil, Air, Water 41, 493–502.
- Noubactep C. (2013b): Metallic iron for water treatment: A critical review. Clean Soil, Air, Water 41, 702–710.
- Noubactep C. (2013c): Metallic iron for environmental remediation: the long walk to evidence. Corros. Rev. 31, 51–59.
- Noubactep C. (2014a): Flaws in the design of Fe(0)-based filtration systems? Chemosphere 117, 104–107.
- Noubactep C. (2014b): Water remediation by metallic iron: Much ado about nothing As profitless as water in a sieve? CLEAN Soil, Air, Water 42, 1177–1178
- Noubactep C. (2015): Designing metallic iron packed-beds for water treatment: A critical review. Clean Soil, Air, Water, doi:10.1002/clen.201400304.
- O'Hannesin S.F., Gillham R.W. (1998): Long-term performance of an in situ "iron wall" for remediation of VOCs. Ground Water 36, 164–170.
- O'Carroll D., Sleep B., Krol M., Boparai H., Kocur C. (2012): Nanoscale zero valent iron and bimetallic particles for contaminated site remediation. Adv. Water Resour. 51, 104–122.
- Obanijesu E.O., Pareek V., Gubner R., Tade M.O. (2010): Corrosion education as a tool for the survival of natural gas industry. Nafta Sci. J. 61, 541–554.
- Odziemkowski M. (2009): Spectroscopic studies and reactions of corrosion products at surfaces and electrodes. Spectrosc. Prop. Inorg. Organomet. Compd. 40, 385–450.
- Odziemkowski M.S., Schuhmacher T.T., Gillham R.W., Reardon E.J. (1998): Mechanism of oxide film formation on iron in simulating groundwater solutions: raman spectroscopic studies. Corros. Sci. 40, 371–389.
- Oh S.-Y. (2009): Factors affecting the reductive transformation of nitroaromatics by zerovalent iron: Dissolved anions, adsorption, iron loss, and possibility of clogging. Geosystem Eng. 12, 27–34.
- Oh S.-Y., Shin D.-S. (2013): Degradation of spent caustic by Fenton and persulfate oxidation with zero-valent iron, J. Chem. Technol. Biot. 88, 145–152.
- Painter B.D.M. (2004): Reactive barriers: Hydraulic performance and design enhancements. Ground Water 42, 609–619.
- Painter B.D.M. (2005): Optimization of permeable reactive barriers for the remediation of contaminated groundwater. Ph.D. Dissertation Natural Resource Engineering Department, Lincoln University, Canterbury, New Zealand.
- Painter B.D.M., Milke M.W. (2007): Comparison of factorial and scenario analysis methods for assessing uncertainty in the design of permeable reactive barriers. Ground Water Monit. Remed. 27, 102–110.

- Pauwels B., Verstraete W. (2006): The treatment of hospital wastewater: an appraisal. J. Water Health 4, 405–416.
- Petrovic M., Hernando M.D., Díaz-Cruz M.S., Barceló D., (2005): Liquid chromatographytandem mass spectrometry for the analysis of pharmaceutical residues in environmental samples: a review. J. Chromatogr. A 1067, 1–14.
- Pham A.L.T., Doyle F.M., Sedlak D.L, (2009): A Silica Containing Iron Oxide Catalyst Capable of Activating Hydrogen Peroxide at Neutral pH Values. Environ. Sci. Technol. 43, 8930–8935.
- Pilling N.B., Bedworth R.E. (1923): The oxidation of metals at high temperatures. J. Inst. Metals, 29 529–591.
- Putra E.K., Pranowo R., Sunarso J., Indraswati N., Ismadji S., (2009): Performance of activated carbon and bentonite for adsorption of amoxicillin from wastewater: mechanism, isotherms and kinetics. Water Res. 43, 2419–2430.
- Radjenovic J., Petrovic M., Ventura F., Barceló D. (2008): Rejection of pharmaceuticals in nanofiltration and reverse osmosis membrane drinking water treatment. Water Res. 42, 3601–3610.
- Rahman M.A., Karmakar S., Salama H., Gactha-Bandjun N., Btatkeu K. B.D., Noubactep C. (2013): Optimising the design of Fe⁰-based filtration systems for water treatment: The suitability of porous iron composites. J. Appl. Solution Chem. Modeling 2, 165–177.
- Reardon J.E. (1995): Anaerobic corrosion of granular iron: Measurement and interpretation of hydrogen evolution rates. Environ. Sci. Technol. 29, 2936–2945.
- Reardon E.J. (2005): Zerovalent irons: Styles of corrosion and inorganic control on hydrogen pressure buildup. Environ. Sci. Tchnol. 39, 7311–7317.
- Richardson J.P., Nicklow J.W. (2002): In situ permeable reactive barriers for groundwater contamination. Soil and Sediment Contamination 11, 241–268.
- Rodgers J.D., Bunce N.J. (2001): Review paper: treatment methods for the remediation of nitroaromatic explosives. Water Res. 35, 2101–2111.
- Rodriguez S., Santos A., Romero A., Vicente F. (2012): Kinetic of oxidation and mineralization of priority and emerging pollutants by activated persulfate. Chem. Eng. J. 213, 225–234.
- Rosenblatt-Farrel, N. (2009): The landscape of antibiotic resistance. Environ. Health Persp. 117, 245–250.
- Roshani B., Karpel Vel Leitner, N. (2011): Effect of persulfate on the oxidation of benzotriazole and humic acid by e-beam irradiation. J. Hazard. Mater. 190, 403–408.
- Saha B., Das S., Saikia J., Das G. (2011): Preferential and enhanced adsorption of different dyes on iron oxide nanoparticles: A Comparative Study. J. Phys. Chem. C 115, 8024–8033.
- Salter-Blanc A.J., Tratnyek P.G. (2011): Effects of solution chemistry on the dechlorination of 1,2,3-trichloropropane by zero-valent zinc. Environ. Sci. Technol. 45, 4073–4079.
- Sarin P, Snoeyink V.L., Bebee J., Kriven W.M., Clement J.A. (2001): Physico-chemical characteristics of corrosion scales in old iron pipes. Wat. Res. 35, 2961–2969.
- Sarin P., Snoeyink V.L., Bebee J., Jim K.K., Beckett M.A., Kriven W.M., Clement J.A. (2004a): Iron release from corroded iron pipes in drinking water distribution systems: Effect of dissolved oxygen. Water Res. 38, 1259–1269.

- Sarin P., Snoeyink V.L., Lytle D.A., Kriven W.M. (2004b): Iron corrosion scales: Model for scale growth, iron release, and colored water formation. J. Envir. Eng. 130, 364–373.
- Sato N. (2000): Electrode potential and local cell. Corros. Rev. 18, 377–394.
- Sato N. (2001): Surface oxides affecting metallic corrosion. Corros. Rev. 19, 253–272.
- Schaffer M., Boxberger N., Börnick H., Licha T., Worch E. (2012a): Sorption influenced transport of ionizable pharmaceuticals onto a natural sandy aquifer sediment at different pH. Chemosphere 87, 513–520.
- Schaffer M., Börnick H., Nödler K., Licha T., Worch E. (2012b): Role of cation exchange processes on the sorption influenced transport of cationic b-blockers in aquifer sediments. Water Res. 46, 5472–5482.
- Shafiquzzaman Md., Azam Md.S., Nakajima J., Bari Q.H. (2011): Investigation of arsenic removal performance by a simple iron removal ceramic filter in rural households of Bangladesh. Desalination 265, 60–66.
- Scherer M.M., Richter S., Valentine R.L., Alvarez P.J.J. (2000): Chemistry and microbiology of permeable reactive barriers for in situ groundwater clean-up. Crit. Rev. Env. Sci. Tec. 30, 363–411.
- Schindelholz E., Kelly R.G. (2012): Werring phenomena and time of wettness in at atmospheric corrosion: a review. Corros. Rev. 30, 135–170.
- Schwartz T., Kohnen W., Jansen B., Obst U. (2003): Detection of antibiotic-resistant 493 bacteria and their resistance genes in wastewater, surface water, and drinking water 494 biofilms. FEMS Microbiol. Ecol. 43, 325–335.
- Scott T.B., Popescu I.C., Crane R.A., Noubactep C. (2011): Nano-scale metallic iron for the treatment of solutions containing multiple inorganic contaminants. J. Hazard. Mater. 186, 280–287.
- Sedlak D.L., Andren A.W. (1991): Oxidation of chlorobenzene with Fenton reagent. Env. Sci. Technol. 25, 777–782.
- Segura Y., Martínez F., Melero J.A. (2013): Effective pharmaceutical wastewater degradation by Fenton oxidation with zero-valent iron. Appl. Catal. B: Environ. 136-137, 64–69.
- Shannon M.A., Bohn P.W., Elimelech M., Georgiadis J.G., Marinas B.J., Mayes A.M. (2008): Science and technology for water purification in the coming decades Nature 452, 301– 310.
- Sharma V.K. (2008): Oxidative transformations of environmental pharmaceuticals by Cl₂, ClO₂, O₃, and Fe (VI): kinetics assessment. Chemosphere 73, 1379–1386.
- Sharp J.H. (1973): Total organic carbon in seawater comparison of measurements using persulfate oxidation and high temperature combustion. Mar. Chem. 1, 211–229.
- Shi C., Wei J., Jin Y., Kniel K.E., Chiu P.C. (2012): Removal of viruses and bacteriophages from drinking water using zero-valent iron. Sep. Purif. Technol. 84, 72–78.
- Shih Y.-J., Li Y.-C., Huang Y.-H. (2013): Application of UV/persulfate oxidation process for mineralization of 2,2,3,3-tetrafluoro-1-propanol. J. Taiwan Inst. Chem. E. 44, 287–290.
- Sikora E., Macdonald D.D. (2000): The passivity of iron in the presence of ethylenediaminetetraacetic acid I. General electrochemical behavior. J. Electrochem. Soc. 147, 4087–4092.
- Snoeyink V.L., Jenkins D. (1980): Water Chemistry. New York: John Willey.

- Stieber M., Putschew A. Jekel M. (2011): Treatment of pharmaceuticals and diagnostic agents using zero-valent iron - kinetic studies and assessment of transformation products assay. Environ. Sci. Technol. 45, 4944–4950.
- Stieber M., Putschew A., Jekel M. (2008): Reductive dehalogenation of iopromide by zerovalent iron. Water Sci. Technol. 57, 1969–1975.
- Stratmann M., Müller J. (1994): The mechanism of the oxygen reduction on rust-covered metal substrates. Corros. Sci. 36, 327–359.
- Suslick K.S. (1989): The chemical effects of ultrasound. Sci. Am. 260, 80-86.
- Svitelska G.V., Gallios G.P., Zouboulis A.I. (2004): Sonochemical decomposition of natural polyphenolic compound (condensed tannin). Chemosphere 56, 981–987.
- Tan C., Gao N., Deng Y, An N., Deng J. (2012): Heat-activated persulfate oxidation of diuron in water. Chem. Eng. J. 203, 294–300.
- Tanboonchuy V., Grisdanurak N., Liao C.-H. (2012): Nitrate probe for quantifying reducing power of nanoscale zero-valent iron. Sustain. Environ. Res. 22, 185–191.
- Tang W.Z., Huang C.P., (1992): Photocatalyzed oxidation pathways of 2,4-dichlorophenol by cds in basic and acidic aqueous-solutions. Water. Res. 29, 745–756.
- Tepong-Tsindé R., Crane R., Noubactep C., Nassi A., Ruppert H. (2015a): Fundamentals for designing sustainable metallic iron based filters for water treatment. Water 7, 868–897.
- Tepong-Tsindé R., Phukan M., Nassi A., Noubactep C., Ruppert H. (2015b): Validating the efficiency of the MB discoloration method for the characterization of Fe⁰/H₂O systems using accelerated corrosion by chloride ions. Chem. Eng. J., doi: 10.1016/j.cej.2015.04.129.
- Ternes T.A. (1998): Occurrence of drugs in German sewage treatment plants and rivers, Water Res. 32, 3245–3260.
- Ternes T.A., Hirsch R. (2000): Occurrence and behaviour of iodinated contrast media in sewage facilities and the aquatic environment. Environ. Sci. Technol. 34, 2741–2748.
- Ternes T.A., Joss A., Siegrist H. (2004): Peer reviewed: Scrutinizing pharmaceuticals and personal care products in wastewater treatment. Environ. Sci. Technol. 38, 392A–399A.
- Thompson L.H., Doraiswamy L.K. (1999): Sonochemistry: science and engineering. Ind. Eng. Chem. Res. 38, 1215–1249.
- Togue-Kamga F., Btatkeu B.D., Noubactep C., Woafo P. (2012): Metallic iron for environmental remediation: Back to textbooks. Fresenius Environ. Bull. 21, 1992–1997.
- Tratnyek P.G., Salter A.J. (2010): Response to Comment on "Degradation of 1,2,3-Trichloropropane (TCP): Hydrolysis, Elimination, and Reduction by Iron and Zinc". Environ. Sci. Technol. 44, 3198–3199.
- Tratnyek P.G., Scherer M.M., Johnson T.J., Matheson L.J. (2003): Permeable reactive barriers of iron and other zero-valent metals. In Chemical Degradation Methods for Wastes and Pollutants: Environmental and Industrial Applications, Tarr, M.A., Ed., Marcel Dekker: New York, 371–421.
- Tsitonaki A., Petri B., Crimi M., Mosbæk H., Siegrist R.L., Bjerg P.L. (2010): In situ chemical oxidation of contaminated soil and groundwater using persulfate: A review. Crit. Rev. Env. Sci. Tec. 40, 55–91.

- Üstün G.E., Solmaz S.K.A., Birgül A. (2007): Regeneration of industrial district wastewater using a combination of Fenton process and ion exchange **e** a case study. Resour. Conserv. Recycl. 52, 425–440.
- Van Orden A.C. (1989): Corrosion mechanisms relevant to high-level waste repositories. Eng. Geol. 26, 331–349.
- Venny Gan S., Ng H.K. (2012): Inorganic chelated modified-Fenton treatment of polycyclic aromatic hydrocarbon (PAH)-contaminated soils. Chem. Eng. J. 180, 1–8.
- Vernon W.H.J. (1945): Controlling factors in atmospheric and immersed corrosion. J. Sci. Instrum. 22, 226–230.
- Vicente F., Santos A., Romero A., Rodriguez S. (2011): Kinetic study of diuron oxidation and mineralization by persulphate: Effects of temperature, oxidant concentration and iron dosage method. Chem. Eng. J. 170, 127–135.
- Waldemer R.H., Tratnyek P.G., Johnson R.L., Nurmi J.T. (2007): Oxidation of chlorinated ethenes by heat-activated persulfate: kinetics and products. Environ. Sci. Technol. 41, 1010–1015.
- Walker K.R., Clifford R.H. (2002): A comparison of three TOC methods on a single analyzer. Am. Lab. 34, 23–24.
- Warner S.D., Sorel D. (2003): Ten years of permeable reactive barriers: Lessons learned and future expectations. In: Chlorinated Solvent and DNAPL Remediation: Innovative Strategies for Subsurface Cleanup, Henry, S.M., Warner, S.D., Eds, American Chemical Society: Washington, DC, ACS Symp., Ser. 837, 36–50.
- Weber E.J. (1996): Iron-mediated reductive transformations: investigation of reaction mechanism. Environ. Sci. Technol. 30, 716–719.
- Werner J. (1951): Amination by Reduction. Ind. Eng. Chem. 43, 1917–1919.
- Werner J. (1961): Amination by reduction. Unit Processes Review. Ind. Eng. Chem. 53, 77–78.
- Westerhoff P., James J. (2003): Nitrate removal in zero-valent iron packed columns. Water Res. 37, 1818–1830.
- Westerhoff P., Yoon Y., Snyder S., Wert E. (2005): Fate of endocrine disruptor, pharmaceutical, and personal care product chemicals during simulated drinking water treatment processes. Environ. Sci. Technol. 39, 6649–6663.
- Westholm L.J., Repo E., Sillanpää M. (2014): Filter materials for metal removal from mine drainage a review. Environ. Sci. Pollut. Res. 21, 9109–9128.
- White A.F., Paterson M.L. (1996): Reduction of aqueous transition metal species on the surface of Fe(II)-containing oxides. Geochim. Cosmochim. Acta 60, 3799–3814.
- Wilson E.R. (1923): The mechanism of the corrosion of iron and steel in natural waters and the calculation of specific rates of corrosion. Ind. Eng. Chem. 15, 127–133.
- Yan J., Zhu L., Luo Z., Huang Y., Tang H., Chen M. (2013): Oxidative decomposition of organic pollutants by using persulfate with ferrous hydroxide colloids as efficient heterogeneous activator. Sep. Purif. Technol. 106, 8–14.
- Yang S., Wang P., Yang X., Wei G., Zhang W., Shan L. (2009): A novel advanced oxidation process to degrade organic pollutants in wastewater: Microwave-activated persulfate oxidation. J. Environ. Sci. 21, 1175–1180.

- Yoon S.-H., Jeong S., Lee S. (2012): Oxidation of bisphenol A by UV/S₂O₈²⁻: Comparison with UV/H₂O₂. Environ. Technol. 33, 123–128.
- You Y., Han J., Chiu P.C., Jin Y. (2005): Removal and inactivation of waterborne viruses using zerovalent iron. Environ. Sci. Technol. 39, 9263–9269.
- Yu J., Liu W., Zeng A., Guan B., Xu X. (2013): Effect of SO₄²⁻ on 1,1,1-Trichloroethane degradation by Fe⁰ in aqueous solution. Groundwater 51, 286–292.
- Zafarani H.R., Gactha-Bandjun N., Tsamo C., Bahrololoom M.E., Noubactep C., Tashkhourian J. (2014): Evaluating the suitability of the testing procedures for alternative adsorbing materials for wastewater treatment. Fresenius Environ. Bull. 26, 1433–1437.
- Zhang J., Chang V.W.C., Giannis A., Wang J.-Y. (2013): Removal of cytostatic drugs from aquatic environment: A review. Sci. Tot. Environ. 445-446, 281–298.
- Zhao D., Liao X., Yan X., Huling S.G., Chai T., Tao H. (2013): Effect and mechanism of persulfate activated by different methods for PAHs removal in soil. J. Hazard. Mater. 254-255, 228–235.
- Zhu L., Ai Z., Ho W., Zhang L. (2013): Core-shell Fe-Fe₂O₃ nanostructures as effective persulfate activator for degradation of methyl orange. Sep. Purif. Technol. 108, 159–165.
- Ziylan A., Koltypin Y., Gedanken A., Ince N.H. (2013): More on sonolytic and sonocatalytic decomposition of Diclofenac using zero-valent iron. Ultrason. Sonochem. 20, 580–586.

9. Appended articles

The present work is based on the following papers (not included in the online version):

- P1. <u>Ghauch, A.</u> (2008) Rapid Removal of Flutriafol from water by zero valent iron powder. Chemosphere 71, 816-826.
- P2. <u>Ghauch, A</u>., Tuqan A. (2008) Catalytic Degradation of Chlorothalonil in water using Bimetallic Iron-Based Systems. Chemosphere 73, 751-759.
- P3. <u>Ghauch, A</u>. (2008) Discussion of Chicgoua Noubactep on "Removal of thiobencarb in aqueous solution by zero valent iron" by Md. Nurul Amin et al. [Chemosphere 70 (2008) 511-515]. Chemosphere 72, 328-331.
- P4. <u>Ghauch, A.</u>, Tuqan, A. (2009) Reductive destruction and decontamination of aqueous solutions of chlorinated antimicrobial agent using bimetallic systems. J. Hazard. Mater. 169, 665-674.
- P5. <u>Ghauch, A.</u> Tuqan, A. Abou Assi, H. (2009) Antibiotic removal from water: Elimination of amoxicillin and ampicillin by microscale and nanoscale iron particles. Environ. Pollut. 157, 1626-1635.
- P6. <u>Ghauch, A.</u> (2010) Response to Dr. Frańska's comments on the paper "Antibiotic removal from water: Elimination of amoxicillin and ampicillin by microscale and nanoscale iron particles." Ghauch et al. (2009) Environ. Pollut. 157, 1626-1635. Environ. Pollut. 158, 3030-3031.
- P7. <u>Ghauch, A.</u>, Abou Assi, H., Tuqan, A. (2010) Investigating the mechanism of clofibric acid removal in Fe⁰/H₂O systems. J. Hazard. Mater. 176, 48-55.
- P8. <u>Ghauch, A.</u>, Abou Assi, H., Bdeir, S. (2010) Aqueous removal of diclofenac by plated elemental iron: bimetallic systems, J. Hazard. Mater. 182, 64-74.
- P9. <u>Ghauch, A.</u>, Abou Assi, H., Baydoun, H., Tuqan, A., Bejjani, A. (2011) Fe⁰-based trimetallic systems for the removal of aqueous diclofenac: mechanism and kinetics, Chem. Eng. J. 172, 1033-1044.

- P10. <u>Ghauch, A.</u>, Baydoun, H., Tuqan, A., Ayoub, G., Naim, S. (2011) Submicrometric iron particles for the removal of pharmaceuticals in water: Application to □□-lactam antibiotics. Adv. Mater. Res. 324, 485-488.
- P11. ²<u>Ghauch, A.</u>, Baydoun, H., Dermesropian, P. (2011) Degradation of aqueous carbamazepine in ultrasonic/Fe⁰/H₂O₂ systems, **Chem. Eng. J. 172, 18-27.**
- P12. Caré, S. Crane, R., Calabro, P.S. <u>Ghauch, A.</u>, Temgoua, E., Noubactep, C. (2013) Modeling the permeability loss of metallic iron water filtration systems, **Clean-Soil**, Air, Water 41, 275–282.
- P13. <u>Ghauch, A.,</u> Tuqan, A.M. (2012) Oxidation of bisoprolol in heated persulfate/H₂O systems: kinetics and products, **Chem. Eng. J. 183, 162-171.**
- P14. <u>Ghauch, A.,</u> Tuqan, A.M., Kibbi, N. (2012) Ibuprofen removal by heated persulfate in aqueous solution: a kinetics study. **Chem. Eng. J. 197, 483-492.**
- P15. <u>Ghauch, A.,</u> Tuqan, A.M., Kibbi, N. Geryes, S. (2012) Methylene Blue discoloration by heated persulfate in aqueous solution. **Chem. Eng. J. 213, 259-271.**
- P16. ³Ghauch, A., Ayoub, G. Naim, S. (2013) Degradation of sulfamethoxazole by persulfate assisted micrometric Fe⁰ in aqueous solution. Chem. Eng. J. 228, 1168-1181.

 $^{^{2}}$ This paper was elected featured article and selected for the cover page of Volume 172 (Issue of August 2011).

³ This paper was elected featured article and selected for the cover page of Volume 228 (Issue of July 2013).